

NOTA DI LAVORO 144.2010

The Public Management of Risk: Separating Ex Ante and Ex Post Monitors

By **Yolande Hiriart**, Toulouse School of Economics (IDEI-LERNA).

David Martimort, Toulouse School of Economics (EHESS, IDEI-GREMAQ)

Jerome Pouyet, Paris School of Economics

INSTITUTIONS AND MARKETS Series Editor: Fausto Panunzi

The Public Management of Risk: Separating Ex Ante and Ex Post Monitors

By Yolande Hiriart, Toulouse School of Economics (IDEI-LERNA). David Martimort, Toulouse School of Economics (EHESS, IDEI-GREMAQ) Jerome Pouyet, Paris School of Economics

Summary

When a firm undertakes risky activities, the conflict between social and private incentives to implement safety care requires public intervention which can take the form of both monetary incentives but also ex ante or ex post monitoring, i.e., before or after an accident occurs. We delineate the optimal scope of monitoring depending on whether public monitors are benevolent or corruptible. We show that separating the ex ante and the ex post monitors increases the likelihood of ex post investigation, helps prevent capture and improves welfare.

Keywords: Risk Regulation, Monitoring, Capture, Integration, Separation

JEL Classification: L51, D82

Financial and intellectual support from the French Ministry of Ecology and Sustainable Development are acknowledged. We are grateful to Bernard Caillaud, Giacinta Cestone, David Encaoua, David Ettinger, Denis Gromb, Jean-Pierre Ponssard, Bernard Salani 'e, Wilfried Sand-Zantman, Jean Tirole, participants at various seminars, two referees and one Editor for helpful comments. All errors are ours.

Address for correspondence:

Yolande Hiriart IDEI-LERNA Manufacture des Tabacs Aile Jean-Jacques Laffont, 21 Allée de Brienne 31000 Toulouse France E-mail: yhiriart@cict.fr

The Public Management of Risk: Separating Ex Ante and Ex Post Monitors*

Yolande Hiriart[†] David Martimort[‡] Jerome Pouyet[§]

23rd June 2010

Abstract

When a firm undertakes risky activities, the conflict between social and private incentives to implement safety care requires public intervention which can take the form of both monetary incentives but also ex ante or ex post monitoring, i.e., before or after an accident occurs. We delineate the optimal scope of monitoring depending on whether public monitors are benevolent or corruptible. We show that separating the ex ante and the ex post monitors increases the likelihood of ex post investigation, helps prevent capture and improves welfare.

Keywords. Risk regulation, monitoring, capture, integration and separation.

JEL Classification. L51, D82.

^{*}Financial and intellectual support from the French Ministry of Ecology and Sustainable Development are acknowledged. We are grateful to Bernard Caillaud, Giacinta Cestone, David Encaoua, David Ettinger, Denis Gromb, Jean-Pierre Ponssard, Bernard Salanié, Wilfried Sand-Zantman, Jean Tirole, participants at various seminars, two referees and one Editor for helpful comments. All errors are ours.

[†]Toulouse School of Economics (IDEI-LERNA). Address: IDEI-LERNA, Manufacture des Tabacs, Aile Jean-Jacques Laffont, 21 Allée de Brienne, 31000 Toulouse, France.

[‡]Toulouse School of Economics (EHESS, IDEI-GREMAQ). Address: IDEI, Manufacture des Tabacs, Aile Jean-Jacques Laffont, 21 Allée de Brienne, 31000 Toulouse, France.

[§]Paris School of Economics. Address: 48 boulevard Jourdan, 75014 Paris, France.

1 Introduction

Although much debated and often criticized, the view that our societies are 'at risk'¹ has certainly pushed both scholars and practitioners to reconsider the role of public intervention in the field of risk regulation.² Consumers should be protected against buying defective products, patients against medical malpractice, workers against accidents in the workplace, the environment against major industrial or transportation hazards, etc. In all these circumstances and although risk tolerance may vary, public intervention is called for to control private actors involved in activities that put humans or the environment at risk. As risks spread over the whole spectrum of economic activities, more effort and expertise should be allocated to assess their true impact on society. The adequate design of incentives for key players involved in the management of these risks should be put at the forefront of the public debate.

Maintaining risk at levels which are socially acceptable does require systems of control. This issue has attracted much attention in the public management literature with a strong motivation being to explain the great variety of regimes in risk regulation across fields and countries.³ However, little is known on the design of adequate institutions for risk regulation. Institutions do vary significantly across fields. Casual evidence suggests that, sometimes, administrative agencies are staffed with experts able to assess specific risks and these agencies have strong enforcement powers. A typical example is that of nuclear power plants, which are routinely checked for maintaining safety. In other fields, such as defective products, agencies are more generalist and most of the enforcement power resides with Courts of Law that perform their own investigation in case of prosecution following an accident. Most of the time however, risk regulation involves an intricate combination of both kinds of intervention taking place either ex ante or ex post. Transportation, road and navigation safety, or occupational safety are good illustrations.

One might argue that risk regulation fits into the general grid already available to discuss regulatory policies and institutions for market regulation.⁴ However, there is some value in distinguishing agencies and regulations which are used ex ante, i.e. before any accident occurs, from agencies, Courts of Law and other enforcement devices which may intervene ex post, i.e. after an accident. This time line naturally distinguishes the roles of different public officials involved in risk monitoring.

In this respect, an important but largely unexplored issue is to delineate the optimal scope of ex ante and ex post control in a world plagued with informational constraints. This paper discusses the costs and benefits of splitting the tasks of ex ante and ex post monitoring of a firm, the activity of which generates some risk. This is done in a context where moral hazard on safety care calls for explicit monetary incentives but also for setting up auditing mechanisms to force compliance with safety standards. When

¹Beck (1992).

²For instance, Hood, Rothstein and Baldwin (2001) recognized that "The idea of the 'regulatory state' is that a new institutional and policy style has emerged, in which government's role as a regulator advances (...)".

³Hood, Rothstein and Baldwin (2001) and Power (1997).

⁴Noll (1989) and Baron (1989).

capture of monitors by the very interests they are supposed to control is a concern, separation of monitoring tasks between two independent bodies is preferable.

Overview of the model. Consider a risk-neutral firm which can cause an accident of substantial scale affecting third-parties. This firm undertakes a nonverifiable prevention effort. A high level of effort is socially optimal. Compliance with this standard of due care can be induced through monetary incentives and the threat of random inspections. Such monitoring might either be preventative (ex ante) or only occur ex post, following an accident, and uncovers whether the firm did perform sufficient care. Fines can be imposed if investigation reveals misconduct, but the firm is protected by limited liability. This access to privileged information gives discretion to public officials. The firm may attempt to capture monitors to prevent them from revealing its misbehavior.

Had monitors been non-corruptible, monitoring would unambiguously improve the firm's incentives. Things are different when monitors are corruptible.

Consider first the case where ex ante and ex post monitoring is performed by a single entity. Such an integrated organization opens large opportunities for collusion. The long-term relationship between an integrated monitor and the firm facilitates collusion by expanding the set of contingencies in which bribes can be exchanged. Under integration, the monitor's close contact with the firm significantly reduces the transaction costs of side-contracting. Postulating convex transaction costs in side-transfers, average transaction costs decrease as bribes are spread over more contingencies.⁵ Intervening both ex ante and ex post, an integrated agency reduces such transaction costs and reaches more efficient collusive deals with the regulated firm.

Under separation, different monitors are used ex ante and ex post. Each monitor anticipates that the other receives enough benefits from adopting an uncorrupted behavior. When striking their collusive deal, the firm and the ex ante monitor anticipate that another monitor may intervene ex post to unveil both the firm's misconduct and evidence of corruptible deals. Bribes can only be transferred when they cannot be detected by an ex post investigation. Smoothing bribes with the ex ante monitor becomes harder and transaction costs of collusion increase. Diseconomies of scale in side-contracting appear, making it easier to prevent capture.⁶

Because it reduces the social cost of preventing capture, ex post monitoring takes place more often under separation than under integration. As its capture is less likely under separation, the ex ante monitor is called upon more often. This highlights a complementarity between ex ante and ex post monitoring. Taking a broader perspective, tougher ex ante regulation and ex post judicial prosecution should come together.

Practical relevance. Our analysis sheds some light on a number of recent institutional changes.

⁵Transaction costs are convex when it is increasingly harder to transfer larger bribes or when such illegal side-transfers can be detected and punished at an increasingly higher rate.

⁶Faure-Grimaud, Laffont and Martimort (2002) also deal with the consequences of ex ante and ex post collusion, but they address different issues.

Transportation. Air transportation offers an illustration of primary interest for our analysis. The investigation of airplane transportation accidents in Canada was, for many years, fully carried out by the Department of Transport. It set the safety standards for the industry, operated elements of the system such as airports and air traffic control, licensed the carriers and the crews, and enforced its own regulations. At the same time, it analyzed the safety failures in the industry in which it had such a pervasive presence. The concern about the independence of the regulator from the industry grew so heatedly that the government proceeded with the creation of the Canadian Aviation Safety Board, an independent regulatory body with an accident investigation mandate. In the framework of our model, this institutional reform is a move towards separation.

The management of nuclear wastes. In this sector, a broad set of precautionary activities can be implemented at the firm level to reduce the likelihood of an accident: care during transportation, employee training programs and radiation competency tests, adoption of best-practice for containment and radiation shielding, integrated safety management, etc. The U.S. Department of Energy (D.O.E.) has inherited the task of cleaning up the radioactive refuse from uranium mines, munitions facilities and other sites around the country. Inside D.O.E., until recently, the Office of Environment, Safety and Health (O.E.S.H.) developed safety policy guidance and provided support to D.O.E. sites, while the Office of Security and Safety Performance Assurance (O.S.S.P.A.) conducted safety oversight. The former office mainly ensures that regulated firms comply with the safety standards whereas the latter office mainly investigates accidents. Recently, the D.O.E. has announced the creation of a new office (the Office of Health, Safety and Security) which will undertake enforcement activities previously carried out by both the O.E.S.H. and O.S.S.P.A. This decision is clearly a move towards integration of the tasks of ex ante and ex post monitoring. Our results below cast doubts on the wisdom of such a move.

Literature. The idea that splitting access to privileged information between specialized agencies reduces the costs of capture has previously been investigated by Laffont and Martimort (1998, 1999, thereafter LM). Several important differences exist with the present setting. First, both the ex ante and ex post monitors have access to the same piece of information (albeit at different points in time due to the specific features of risk regulation), namely the agent's level of safety care, whereas separated agencies get access to different pieces of information in LM. Second, another difference comes from the source of the economies of scale in side-contracting. In LM, "Chinese walls" between regulators create asymmetric information in side-contracting. This undermines the efficiency of the side-deals that each regulator reaches with the firm. Here instead, the collusive gains from integration come from the fact that a single regulator can better smooth bribes over the different states of nature.⁷ Third, in LM, the monitoring technologies that give informative signals to the firm's monitors are exogenously given whereas we devote some attention to the impact of different institutional choices

⁷This desire for bribes smoothing to reduce transaction costs of side-contracting comes from their assumed convexity. Faure-Grimaud and Martimort (2003) present another model building on that assumption.

on the endogenous likelihood of each round of investigations.⁸

Other contributions have highlighted the costs and benefits of splitting public bodies. On the benefits side, Kofman and Lawarée (1993) show that bringing an uncorruptible monitor may limit the scope for capturing corruptible ones. Our model instead does not assume a priori that this extra monitor is uncorruptible but derives the benefits of separation in a model where both ex ante and ex post supervisors can be captured. Kofman and Lawarée (1996a) find that competing agencies may be useful in a yardstick model where they acquire correlated signals. In our model, monitors have instead access to different signals which are conditionally independent and such yardstick mechanisms lose appeal. On the costs side, Shleifer and Vishny (1993) study the optimal number of public officials controlling a given firm. Public officials choose noncooperatively the bribes they require. In equilibrium, excessive bribery occurs with several officials. Although such a result might be convincing in weak institutional environments, the stake for bribery in Shleifer and Vishny (1993) is exogenous and no attention is given to possible institutional responses aimed at limiting the wasteful competition between bureaus. Instead, in our model, this stake arises endogenously from asymmetric information. We give a particular attention on the institutional response to the threat of capture, not only in terms of incentives for public bodies, but also in terms of their overall organization.

Finally, the literature on corruption in law enforcement (Becker and Stigler, 1974; Mookherjee and P'ng, 1995; Garoupa, 1997; Polinsky and Shavell, 2001) analyzes the impact of corruption on the likelihood of investigation in various contexts but does not draw, as we do, the implications of corruption for institutional design. Corruption is an equilibrium phenomenon in these models. In our context instead, the Collusion-Proofness Principle⁹ always holds so that institutions are robust to the threat of capture. The best institutional form minimizes the cost of preventing capture. This institutional perspective is also the focus of Boyer and Porrini (2001, 2004) who compare ex ante regulation and various liability rules enacted ex post. They postulate a priori that the legal system viewed as an ex post monitor is immune to capture, whereas we derive this result from equilibrium behavior. In addition, they analyze separately the costs and benefits of the two systems whereas we model their joint use.¹⁰

Section 2 presents our theoretical model. Section 3 studies the benchmark without collusion. Section 4 describes our modeling assumptions for capture and studies the impact of collusion on monitoring under integration and separation. Section 5 performs a welfare comparison between institutional modes and highlights the possible complementarity between ex ante and ex post monitoring that might arise endogenously under separation. Section 6 discusses some implications of our results and presents several extensions. Section 7 concludes. All proofs are in the Appendix.

⁸Last, we focus here on moral hazard as the source of the rent that the firm wants to protect by capturing its monitors, whereas LM deal with adverse selection.

⁹See Tirole (1986) for instance.

¹⁰Earlier contributions, like Wittman (1977), Shavell (1984a, 1984b), Kolstad, Ulen and Johnson (1990), and Mookherjee and P'Ng (1992) compare the use of ex ante regulation and ex post liability rules but impose exogenous constraints on instruments, on information gathering technology, or on both. These institutional issues are instead at the core of our analysis.

2 The Model

Consider a firm running a socially risky technology. The probability of an accident affecting third-parties is reduced when this firm implements some safety care. Moral hazard in choosing this variable calls for controlling whether the firm abides to a standard of due care or not.

2.1 Incentives, Information and Control

Moral hazard. Following an accident, third-parties suffer from a damage of social value D > 0. The probability of such an accident, $1 - \pi(e)$, is decreasing in the firm's effort *e*. For simplicity, effort takes only two values, $e \in \{0, 1\}$, so that these probabilities are $1 - \pi_1 < 1 - \pi_0$ (we denote $\Delta \pi = \pi_1 - \pi_0$). Implementing effort $e = 1 \operatorname{costs} \psi$ to the firm whereas e = 0 is costless.¹¹

Moral hazard stems both from the non-verifiability of effort and from the conflict between social and private incentives to exercise care. Provided that the damage is large, i.e. $D \ge \frac{\psi}{\Delta \pi}$, efficiency calls for implementing the high level of effort; but the firm prefers to save on the compliance cost.

The firm is protected by limited liability and has no further assets that could be seized in case of an environmental accident.^{12,13} However, it has some hidden wealth w > 0 that can be used for bribery purposes if needed.

Contracts. A regulatory incentive scheme stipulates the firm's payments conditional on its environmental performances, i.e., whether an accident has taken place or not. Because of limited liability, the only relevant payment consists of a non-negative incentive reward *t* following a good environmental performance.¹⁴

Although we focus on monetary rewards and punishments, a broader interpretation of regulatory incentive payments should be kept in mind so that our model also fits institutional contexts where regulatory rewards are banned. For instance, industrial accidents sometimes come with damages to the fixed capital of the firm or/and to some stakeholders (e.g., workers). Costs may also be indirect and include tightened future regulations, permit refusals by the government, new taxes or future boycotts by environmentally-oriented consumers. Rewards may also involve the firm's implicit reputational gains vis-à-vis customers, potential contracting partners, the government,

¹¹The cost of effort is non-monetary for simplicity although our modelling could easily be modified to take into account monetary costs without changing our main results.

¹²We could allow for a positive amount of seizable assets as long as it is not too large so that the firm only exercises effort when given some positive liability rent.

¹³Firms running risky activities are generally protected by limits on their liability since the consequences of large scale accidents are so staggering that no insurance companies would fully insure them (see, e.g., the Price-Anderson Act in the U.S. for nuclear activities). Moreover, on top of institutional restrictions, risky ventures often enter into various activities ("flight-by-night" techniques, spin-offs of subsidiaries, ...) whose goal is to hide seizable assets. Pitchford (2001) and Hiriart and Martimort (2006) analyze the issue of extended liability when ex post legal intervention unveils new funds to compensate victims.

¹⁴In full generality, a contract should also stipulate a payment in the event of an accident. However, given the firm's limited liability, this payment would optimally be equal to zero at the optimum.

shareholders and the financial community.¹⁵ Along the same lines, the regulator's choice of incentive fines and rewards could be replaced by his ability to harden or soften future regulations when either canceling or permitting the firm's new products and activities for instance.¹⁶ With these interpretations in mind, it becomes easier to map our findings with existing regulatory practices. For instance, a regulatory agency that only controls safety standards and, a priori, does not directly affect the firm's financial incentives, may implement lax standards in the future as an imperfect substitute for monetary rewards.

Monitoring. Monitoring by public bodies might occur either before or/and after an accident. To gather hard evidence on whether the firm complies with standards, on-site random inspections are often used in the regulation of risky activities. Such investigations take place ex ante, i.e. before any accident ever occurs. Instead, ex post audits are run by an accident investigation commission or by the judiciary system following accidents. Of course, this intervention is relevant only when the ex ante investigation did not take place or failed to release how much effort was undertaken by the firm.

At a cost $C(p_r)$ standing for the cost of resources allocated to ex ante monitoring, an ex ante monitor observes a signal σ_r on the firm's effort level with probability p_r . To focus on interior solutions, $C(\cdot)$ is strictly increasing, sufficiently convex to ensure concavity of the optimization problems below, with C(0) = 0, and the Inada conditions C'(0) = 0, $C'(1) = +\infty$. The signal $\sigma_r \in \{e, \emptyset\}$ either reveals the firm's effort or is uninformative with respective probabilities $\epsilon \in (0, 1)$ and $1 - \epsilon$. The probability ϵ captures the precision of the signal.¹⁷

When its misconduct has been detected by the ex ante monitor, the firm is punished with some fines f_r . These punishments simply consist in suppressing part of the regulatory incentive payment that the firm would have received in case of a good environmental performance, i.e. $f_r \leq t$. Of course, the *Maximal Punishment Principle*¹⁸ applies in our context so that incentives for compliance are fostered when $f_r = t$. This Principle is used throughout to simplify exposition.¹⁹

¹⁵Lesourd and Schilizzi (2001) discuss the various indirect costs and benefits of environmental risks.

¹⁶Hiriart and Martimort (2006) provide more motivation for the short-cut of using monetary transfers between the Agency and the regulated firm.

¹⁷We thus assume that the monitor may either get perfect information on the effort or no informative signal. This assumption is standard in the collusion literature in vertical hierarchies since the seminal paper by Tirole (1986). It ensures that collusion takes place only when monitors have complete information on the agent's effort. If instead the monitor had a noisy signal on the firm's effort, collusion would take place under asymmetric information which might introduce further inefficiency. The corresponding transaction costs due to asymmetric information in side-contracting would induce effects close to those already stressed by LM. Those new effects would superimpose on those we highlight below. We chose thus to follow a simpler modeling path focusing on the novel insights due to the convexity of transaction costs.

¹⁸Quite intuitively, this Principle states that incentive constraints are all the more relaxed that punishments are set at their maximal value in case a noncompliant behavior is detected. See Becker (1968) and Baron and Besanko (1984) for earlier analysis.

¹⁹For the sake of simplicity, we assume that the firm, when found shirking by the ex ante audit, cannot change its effort level to adopt the standard. The implicit assumption here is that such an adoption requires major changes in the production technology and these changes take time. In a previous version of this paper, we showed that the qualitative conclusions are unchanged if the ex ante monitor could

Following an accident, and in case it is still unknown whether the firm has abided to the standard, an ex post investigation might occur with probability p_j . The ex post monitor gets a hard information signal $\sigma_j \in \{e, \emptyset\}$ which again perfectly reveals the firm's effort with probability $\epsilon \in (0, 1)$, or is uninformative. On top of detecting effort, this ex post investigation also unveils how much of the hidden funds w are used by the firm for bribery purposes if the ex ante and ex post monitors are two different bodies. To keep symmetry in modelling the ex ante and ex post rounds of investigation, the administrative cost of an ex post investigation is still given by $C(p_j)$.²⁰ If the ex post monitor figures out that there has been no compliance with the standard of safety care, the firm could a priori be fined. However, remember that the firm's payment is zero following a bad environmental performance so that no fines are actually available for the ex post monitor.

Institutional design. We consider two institutional settings. Under *integration*, the ex ante and ex post monitors are merged into a single entity. Under *separation*, monitors are kept apart and behave non-cooperatively. To fix ideas, one may think of the ex ante monitor as a regulator and the ex post one as a judge. Accordingly, in our analysis, regulators and judges differ along two dimensions: the timing of their intervention and the magnitude of the financial penalties they can impose to the firm. Note, however, that our model admits broader and less specific interpretations that are illustrated by the motivating examples stressed in Section 1.

Monitors' wages. If a monitor does not reveal information on the firm's effort, he gets a base-payment normalized at zero. When he reports evidence about the firm's effort, he may receive a positive wage. Let V_r denote the ex ante monitor's wage when he reports the firm's effort after an ex ante investigation. Let similarly V_j be the ex post monitor's wage.²¹ These wages should be broadly interpreted. They can stand for the share of the agency budget or resources that can be diverted for private use.²² Alternatively, they can also be considered as proxies for career concerns.²³

With probability $p_r \epsilon$, the ex ante intervention succeeds. When it does not and an accident occurs, the ex post monitor is called upon to inspect the firm. When the firm

force the firm to adopt immediately the high standard of safety care when $\sigma_r = 0$.

²⁰Note that the ex ante and ex post technologies are assumed to be identical to simplify exposition. Most of our results are robust to introducing asymmetries in the costs of monitoring ($C_r(.) \neq C_j(.)$) and differences in the probabilities of getting an informative signal at each round of investigation ($\epsilon_r \neq \epsilon_j$).

²¹In full generality, different wages could a priori be offered to the ex post monitor depending on whether an ex ante investigation failed or has not been performed. It turns out that this added degree of generality is not necessary since the optimal collusion-proof wages in those two cases are the same. Therefore, our findings below remain true when this possibility is a priori allowed. Henceforth, our presentation is simplified by assuming directly that those wages are equal.

²²Niskanen (1971) and Laffont and Tirole (1993).

²³If this latter perspective is taken, payments to monitors could simply be viewed as the product of the probability of getting a promotion times the private benefit associated to this new job. Whatever the interpretation behind these wages, they remain socially costly. For instance, rewarding a monitor for a zealous behavior by moving him towards higher positions in the bureaucratic hierarchy may come at the opportunity costs of not rewarding somebody more talented for this job.

complies with the standard, the expected wage left for monitoring is thus defined as:

$$V(p_r, p_j) = p_r \epsilon V_r + (1 - p_r \epsilon)(1 - \pi_1) p_j \epsilon V_j,$$

with

$$V_r, V_j \ge 0,\tag{1}$$

where the latter constraint ensures that monitors get more than their reservation wage.

Social welfare. The regulatory objective²⁴ incorporates the surplus *S* generated by the firm's activity,²⁵ the expected harm on third-parties, the firm's rent, the social cost of regulatory transfers and wages and, finally, the administrative costs of monitoring so that, overall, it writes as:

$$W = S - (1 - \pi_1)D - \pi_1 t + \gamma U - V(p_r, p_j) - C(p_r) - C(p_j) \text{ where } \gamma \in [0, 1)$$
$$\equiv S - (1 - \pi_1)D - \psi - (1 - \gamma)U - V(p_r, p_j) - C(p_r) - C(p_j).$$

Following the tradition of the incentive regulation literature starting with the seminal paper by Baron and Myerson (1982), this objective puts a weight γ less than one on the firm's profit. This ensures that one euro pocketed by the firm is worth less than one euro left elsewhere in the economy. As it can be readily seen on the above expression, this ensures in turn that the firm's rent is socially costly and should be minimized under all contracting circumstances.²⁶

Of course, *S* is large enough so that the firm's activity is valuable, i.e. W > 0 at the optimum. Fines for misconduct should be pocketed by the State. But, since the firm is induced to comply with the standard and no such misconduct is detected at equilibrium, fines do not appear in this expression of social welfare. The existence of these fines nevertheless helps to relax the firm's incentive constraint and to reduce its liability rent.²⁷

We could easily generalize this objective function to take into account redistributive concerns. For instance, the regulatory charter could be more or less aligned with the harmed third-parties, depending on whether the risk at stake is global in nature or

²⁴In what follows, we will sometimes refer to the constitution in charge of designing the overall incentive package as the 'principal'.

²⁵Note that we make the simplifying assumption that the firm's effort does not affect social surplus. It would be a simple extension of our analysis to have the firm's effort also affect social surplus. For instance, we could incorporate a (maybe positive) impact of the firm's effort on social surplus under the form $S + \Delta S \pi_1$. Our analysis would go through by simply changing variables and defining $S' \equiv S - \Delta S$ and $D' \equiv D - \Delta S$.

²⁶We could slightly generalize this expression and have the regulatory (ex ante) and the judicial (ex post) bureaucracy's payoffs also be possibly weighted in the social welfare function. As long as these weights are less than one so that, overall, leaving payoffs above their reservation values to these public bodies is found socially costly, the same analysis as below could be carried over.

²⁷Our formulation of the government's objective is more general than it might appear at first glance. Suppose that the firm makes a profit Π which is taxed for an amount y_a in case of an accident and y_n in case no such accident occurs. Consumer's surplus is s so that our previous formulation applies provided one defines $S = \Pi + s$, $t = \Pi - y_n$ and observes that the binding limited liability constraint of the firm writes then as $y_a = \Pi$.

more local (i.e., whether it affects or not a significant share of the electorate).

Timing. For simplicity, we describe the sequence of events in the case of benevolent monitors. The case of non-benevolent monitors is analyzed in more details in Section 4.1.

• Date 0. The regulatory charter specifies the incentive transfer t, the respective probabilities (p_r, p_j) of ex ante and ex post investigations, fines when misbehavior is detected either ex ante or ex post, and wages for the monitors depending on their reported signals.²⁸

• Date 1. The firm exercises an effort $e \in \{0, 1\}$.

• Date 1⁺. The ex ante monitor inspects the firm with probability p_r . He learns signal σ_r about the firm's effort. If he detects misconduct ($\sigma_r = e = 0$), he imposes the fine $f_r = t$ that applies if no accident takes place later on.

• Date 2. An accident occurs with probability $1 - \pi(e)$.

• Date 2^+ . Following an accident and if nothing has been learned ex ante, the ex post monitor investigates with probability p_i .

• Date 3. Transfers and fines, if any, are paid.

2.2 Incentive Compatibility

To induce the firm to exercise care, the following incentive compatibility constraint must hold:

$$U \equiv \pi_1 t - \psi \ge (1 - p_r \epsilon) \pi_0 t.$$
⁽²⁾

The l.h.s. of (2) is the firm's expected profit if it complies with the standard since, in that case, the firm is never fined when monitoring (either ex ante or ex post) succeeds. The r.h.s. of (2) is the firm's expected profit if it does not comply with the standard. With probability $p_r\epsilon$, an ex ante inspection occurs and misbehavior is detected. The firm is then fined and loses the incentive reward t. With probability $1-p_r\epsilon$, ex ante monitoring either does not take place or it takes place and fails. The transfer t is received if the accident does not occur. If it occurs, ex post monitoring might then be called upon and detect violation with probability $p_j\epsilon$ but, since the firm has no seizable wealth in that state, no fines are imposed.

The incentive constraint (2) yields a lower bound on the incentive payment *t*. Of course, the optimal regulation aims at minimizing the firm's rent and thus at minimizing this incentive reward so that, at the optimum, we have:

$$t^*(p_r,\epsilon) = \frac{\psi}{\Delta \pi + \pi_0 p_r \epsilon}.$$
(3)

There are two ways of providing incentives: through monitoring and through the incentive reward in the event of a good environmental performance. Moreover, as

²⁸We assume that there is full commitment to the regulatory charter. For the implications of limited commitment in an environment with corruptible auditors, see Strausz (1997), Khalil and Lawarée (2006) and Hiriart, Martimort and Pouyet (2010).

shown in the Appendix, the elasticity of the reward with respect to the ex ante audit probability is smaller than one, and decreases when the audit technology becomes less accurate.

Using this expression of the transfer, we may as well obtain the expression of the firm's expected profit as:

$$U = \mathcal{U}(p_r) = \frac{\pi_0(1 - p_r \epsilon)}{\Delta \pi + \pi_0 p_r \epsilon} \psi.$$
 (4)

The firm's limited liability rent $U(\cdot)$ is positive, decreasing and convex in p_r . Such a positive profit is a necessary ingredient to have a stake for capture when monitors are not benevolent, as we will show below.

2.3 Motivation for Contractual Restrictions

One could argue that the space of contracts available to the principal, as described above, is somewhat incomplete. Indeed, the firm's final payments in each state of nature, which are either t or 0, only depend on the occurrence or not of an environmental damage or on whether shirking is detected by a round of monitoring (either ex ante or ex post).

Suppose instead that these payments could also be made contingent on the information revealed by monitoring when it is favorable to the firm. To simplify, let us first consider the information revealed by a round of ex ante monitoring. A simple contract that gives a reward t(1) only when ex ante monitoring reveals that the firm has exercised an effort and there is no accident implements the first-best at no agency cost for the principal. Indeed, it then never pays off for the firm to deviate and exercise zero effort. Therefore, the firm's incentive constraint writes as follows:

$$p_r \epsilon \pi_1 t(1) - \psi \ge 0.$$

This incentive constraint is identical to the firm's participation constraint. At the optimum, this participation constraint is binding and the incentive constraint is automatically satisfied. Agency costs are thus null in that environment.

Intuitively, when the outcome of monitoring is verifiable, monitoring makes effort (indirectly) contractible even though it is only the case with some probability. We are then back to a setting where the non-verifiability of effort does not matter and the firm gets no liability rent for undertaking such effort. In such a setting, the scope for collusion between the firm and its monitors to protect the liability rent, which is our focus throughout the paper, would disappear.

To avoid such an unpalatable conclusion and stress the conditions under which our restricted space of contracts is relevant, suppose that the outcome of monitoring, although hard information, is nonverifiable. Clearly, the extended mechanism described above can be manipulated by the principal himself, by hiding any evidence favorable to the firm whereas such manipulations do not arise when unfavorable evidence is detected.²⁹ Indeed, if the principal observes through a round of ex ante monitoring that

²⁹McLeod (2003) makes a similar point in a simpler model with no monitoring and where the agent's

the firm has implemented some care, claiming that monitoring has not unveiled the firm's effort always saves the expected reward $\pi_1 t(1)$ left to the agent. Instead, had the firm deviated and exercised zero effort, the principal would still have incentives to report that unfavorable information to avoid paying anything to the firm. In other words, assuming that the signal revealed by monitoring is nonverifiable introduces a fundamental asymmetry between "good" and "bad" news on the firm's effort. "Good news" are hidden by the principal if the firm must be rewarded in that state, whereas "bad news" are readily revealed.

Avoiding these manipulations by the principal himself requires paying the same amount to the firm whether monitoring may unveil favorable evidence or not. The only remaining tool to provide incentives to the firm is to offer payments contingent on whether an accident occurs or not, and to suppress these rewards in case monitoring unveils shirking. This argument applies as well to ex post monitoring if any. Overall, the scope for the principal's manipulation requires to pay the firm a fixed payment *t* in case there is no environmental damage, and zero otherwise. This validates our initial focus on what could have been seen as an a priori ad hoc restriction on contracts.

An alternative formulation to avoid that incentive rewards are made contingent on the favorable outcome of an audit is to assume that the incentive payment *t* is not fixed by the 'environmental' principal in charge with organizing the control of safety care but, instead, by another 'economic' principal. This 'economic' principal does not have access to the information on the outcomes of those audits and cannot commit to rewards contingent on those outcomes. He only observes the environmental performances of the firm and whether damages occur or not. On top, the 'environmental' principal has no funds on his own and can certainly impose fines, but not reward the firm following "good" news on the level of safety care it has implemented. This alternative approach is particularly consistent with our interpretation of incentives rewards and payments as reduced-forms for the future benefits of a good environmental performance when direct regulatory rewards are banned.

3 Benevolent Monitors

Let us suppose that the ex ante and ex post monitors are both benevolent. There is no need to pay any positive wage to induce these monitors to reveal informative signals on the firm's effort. In such an environment, separation and integration are clearly equivalent.

Ex ante monitoring punishes the firm since it suppresses rewards following a good environmental performance when misconduct is detected. Ex post monitoring is useless because it does not help to relax the firm's incentive constraint and can only waste administrative resources.

Proposition 1 With benevolent monitors, the optimal probability of ex ante investigation is

performance itself is manipulable by the principal.

strictly positive whereas the probability of ex post investigation is zero:

$$p_r^* > 0$$
 and $p_i^* = 0$.

The optimal incentive payment for the firm is non-negative:

$$t^* > 0.$$

Wages for both rounds of monitoring are zero:

$$V_r^* = V_j^* = 0.$$

4 Preventing Capture

4.1 Collusion Technology

Collusive side-contracts. A firm might bribe its monitors so that they hide information on misconduct. Bribes might take the form of promises for future job opportunities in the private sector for current regulators, direct monetary bribes or campaign contributions targeted towards lawmakers and key elected officials who have influence at the various stages of the firm's monitoring.

As already mentioned, the firm has at its disposal some hidden wealth w > 0 for bribing its monitors.³⁰ The firm has all bargaining power in proposing bribes to the monitor(s).³¹ A corruptible monitor accepts bribes if he gets more by doing so and being lenient than by remaining honest and releasing misconduct that he may have observed through investigation. The ex ante monitor's discretion comes from his ability to report having observed $\sigma_r = \emptyset$ when indeed $\sigma_r = e = 0$. The firm avoids then paying fines. This collusive strategy may be attractive if the monitor gets a share of the corresponding gains pocketed by the firm. The ex post monitor could a priori also enjoy similar benefits by hiding ex post evidence of misconduct, i.e., reporting $\sigma_j = \emptyset$ when $\sigma_j = e = 0$.

The firm commits to offer the promised bribes to the monitor if he remains lenient. Reciprocally, the monitor commits to stay silent on misconduct.³²

Monitors only enjoy private benefit $k(\tau) \in [0, \tau]$ when they receive a bribe $\tau \ge 0$ (in particular, k(0) = 0). Such frictions are due to the existing transaction costs of side-contracting, including possibly the cost that firms may bear in organizing corruptible activities. The existence of such transaction costs is a standard assumption in the public choice and regulation literatures (see Congleton, 1984, and Faure-Grimaud

 $^{^{30}}$ One can easily show that the regulatory agency cannot induce the firm to reveal w through any incentive mechanism.

³¹This assumption simplifies presentation but, given that collusion takes place under complete information between the firm and its monitor(s), nothing would be changed by considering alternative allocations of the bargaining power. Indeed, collusion is valuable whenever these parties get a greater collective surplus than by behaving.

³²See Vafai (2002) for another model of collusion in a moral hazard environment with a risk-averse agent based on similar enforceability assumptions.

and Martimort, 2003, among others). Importantly, the function $k(\cdot)$ is positive, increasing, and strictly concave. Convexity of the transaction costs $TC(\tau) = \tau - k(\tau)$ of side-contracting reflects the fact that collusive partners find it increasingly harder at the margin to transfer greater bribes. Colluding partners look for side-deals that minimize the dead-weight loss associated to these transaction costs.

Given the importance of our assumption on the convexity of transaction costs for our results, it is important to give it some background. Possible justifications behind this assumption may be found both on the demand and the supply sides of the market for bribes. On the demand side, first, we may assume that larger bribes are (exogenously) publicly detected with an increasingly higher probability $P(\tau)$, and that such detection inflicts a large (non-modeled) non-monetary punishments on violators. In this case, we have $k(\tau) = \tau(1 - P(\tau))$, $k'(\tau) = -\tau P'(\tau) + 1 - P(\tau)$, the latter being positive as long as $\tau \leq \tau^* = \frac{1 - P(\tau^*)}{P'(\tau^*)}$, ³³ and $k''(\tau) = -2P'(\tau) - \tau P''(\tau) \leq 0$ when $P''(\tau) \geq 0$.

On the supply side, these transaction costs of side-contracting capture all the frictions that the firm may face in organizing its corruptive activities. With that alternative interpretation in mind, convexity simply means that it becomes harder at the margin to siphon funds of the general budget of the firm for hidden illegal activities, an assumption that seems reasonable.

Finally, we stress that the technology for side-contracting is kept constant as one changes organizational mode. This allows us to focus on the sole role of the difference in the costs of preventing capture across organizations as a determinant of organizational choices.

Last, when investigating the organization with separation, we rule out the possibilities that monitors directly collude one with the other. This can be justified when the transaction costs of such side-contracts are prohibitively high, or when such collusion between public bodies would easily be detected.

Collusion-proofness. The possibility that the firm may enter into side-deals with monitors can be viewed as a new dimension of moral hazard.³⁴ Since side-contracts are non-verifiable by definition and thus cannot be directly banned, the regulatory charter must not only induce a high level of safety care, but also prevent the firm from offering collusive side-contracts.

When damages are large enough, restricting attention to collusion-proof regulatory policies is clearly optimal. It is akin to ensuring that the firm does not want to simultaneously deviate by exercising a low level of care and colluding with monitors.³⁵

³³Note that such τ^* is uniquely defined when the monotone hazard rate property $\frac{d}{d\tau} \left(\frac{1-P(\tau)}{P'(\tau)} \right) < 0$ holds.

³⁴Laffont (1990) pointed out this analogy between collusion and moral hazard.

³⁵The reader knowledgeable in the theory of collusion may recognize that this restriction to the collusion-proof regulatory policies is not as standard as in the literature. The latter has shown that there is no loss of generality in restricting the analysis to those incentive mechanisms that come unchanged through collusion. The bulk of the argument is that any other contracting outcome, where collusion is allowed and played on a continuation equilibrium following a given offer by the principal, yields payoffs that could as well have been reached through such a collusion-proof mechanism. This can be true of the optimal payoff, as in the literature on collusion with hard information (Tirole 1986,

4.2 Integration

Suppose first that both monitors are merged into a single entity. The latter's discretion comes from his ability to coordinate reports at the various stages of investigation. When collusion starts ex ante, an enforceable side-contract gives the colluding partners the opportunity to specify bribes following ex post investigation if any is called upon.

Ex post collusion. Let us begin by the simple case of an ex post collusion. If an ex ante investigation fails or is not carried out, the monitor can still intervene ex post. Ex post collusion might occur if the firm is found negligent and the monitor obtains a signal σ_j which is informative on the firm's misconduct.

With such an ex post collusion, a bribe $\tau_j = k^{-1}(V_j)$ is enough to leave the monitor just indifferent between colluding or not ex post. Remember now that the firm's net regulatory transfer following an accident is zero. Colluding secures payoff $w - k^{-1}(V_j)$ whereas not colluding, i.e. paying no bribes, yields w.

Therefore, ex post collusion does not take place as soon as the following *ex post collusion-proofness constraint* holds:

$$V_j \ge 0. \tag{5}$$

Ex ante collusion. When an ex ante investigation succeeds, the colluding monitor hides information on the low effort undertaken by the firm. The monitor commits not to reveal information if an ex post investigation takes place and also generates hard evidence on the firm's misconduct. In exchange, the firm offers a sequence of bribes $(\tau_r^0, \tau_r^1, \tau_r^2)$. These bribes are respectively offered when an accident does not occur $(\tau_r^0$ with probability π_0), when an accident occurs but ex post investigation fails $(\tau_r^1$ with probability $(1 - \pi_0)(1 - p_j\epsilon))$, and when an accident occurs and ex post investigation succeeds $(\tau_r^2$ with probability $(1 - \pi_0)p_j\epsilon$).

The optimal side-contract $(\tau_r^0, \tau_r^1, \tau_r^2)$ thus solves:

$$(\mathcal{P}^{I}): \max_{\tau_{r}^{0}, \tau_{r}^{1}, \tau_{r}^{2} \in [0,w]} w + \pi_{0}(t - \tau_{r}^{0}) - (1 - \pi_{0})(1 - p_{j}\epsilon)\tau_{r}^{1} - (1 - \pi_{0})p_{j}\epsilon\tau_{r}^{2}$$

subject to

$$\pi_0 k(\tau_r^0) + (1 - \pi_0)(1 - p_j \epsilon) k(\tau_r^1) + (1 - \pi_0) p_j \epsilon k(\tau_r^2) \ge \max\left\{V_r; (1 - \pi_0) p_j \epsilon V_j\right\}.$$
 (6)

To collude, the monitor's gain from accepting the side-contract must exceed his wage if he refuses it. Absent collusion, the monitor may either report evidence early and earn V_r , or hide evidence and wait until he is called upon ex post and learns a new signal. His expected earning is then $(1 - \pi_0)p_j \epsilon V_j$.

^{1992),} or this can be true of any allocations (even suboptimal ones), as when collusion takes place under soft information (Laffont and Martimort 1997, 2000). Here instead, we do not show such general equivalence. Instead, we just posit that any scheme that might induce collusion generates an accident with a higher probability. Our assumption of a large damage *D* implies that welfare is strictly lower with such a mechanism.

We consider parameter values so that the monitors' wages satisfy:

$$V_r \ge (1 - \pi_0) p_j \epsilon V_j. \tag{7}$$

This condition will be checked at the optimum.³⁶ In that case, the best strategy for the monitor when not colluding is to report early the firm's misconduct.

Because transaction costs are convex, smoothing bribes and offering a same bribe τ_r^I under all contingencies reduces the dead-weight loss of side-contracting. Given that the firm has all bargaining power when designing side-contracts, it optimally offers a flat bribe that makes the monitor just indifferent between colluding or not:

$$\tau_r^0 = \tau_r^1 = \tau_r^2 \equiv \tau_r^I = k^{-1} (V_r).$$

Such a bribe is feasible when the firm has enough hidden wealth available, i.e., $w > \tau_r^I$ or, equivalently:

$$V_r < k(w). \tag{8}$$

With such a flat bribe, the firm's overall payoff from ex ante collusion is thus $w + \pi_0 t - k^{-1}(V_r)$. Instead, if the firm does not offer the bribe τ_r^I and misconduct is revealed, the firm keeps w. A regulatory scheme is *ex ante collusion-proof* when the firm prefers not to collude, i.e., when the wage V_r satisfies the following *ex ante collusion-proofness constraint*:

$$V_r \ge k \left(\pi_0 t\right). \tag{9}$$

Incentive compatibility and collusion-proofness. Let us now describe the firm's incentive constraint when collusion matters. As already stressed, the regulatory charter must not only induce the firm to comply with the standard but also not to collude with its monitor, be it either ex ante or ex post. The following *generalized incentive constraint* aggregates all possible deviations available to the firm:

$$U \ge p_r \epsilon \max\left\{0; \pi_0 t - k^{-1}(V_r)\right\} + (1 - p_r \epsilon) \left(\pi_0 t + (1 - \pi_0) p_j \epsilon \max\left\{0; -k^{-1}(V_j)\right\}\right).$$
(10)

Clearly, in the state where an accident takes place and ex post investigation unveils misbehavior, paying a bribe is useless since the firm has nothing to save.

Response to the threat of capture. Since giving up some extra rents to the monitor is socially costly, the collusion-proofness constraints (5) and (9) are both binding at the optimum. Avoiding capture requires to give up the following wages:

$$V_r^I = k(\pi_0 t)$$
 and $V_i^I = 0.$ (11)

With these wages, the firm's incentive constraint remains the same with or without collusion, i.e. (10) coincides with (2). Hence, for fixed investigation probabilities, the

³⁶The monitor's strategy consisting in waiting for some evidence expost is also a possibility on the equilibrium path, i.e., when the firm exercises an effort. The corresponding constraint, $V_r \ge (1 - \pi_1)p_j \epsilon V_j$, holds a fortiori since $\pi_1 > \pi_0$.

firm receives the same liability rent $\mathcal{U}(p_r)$ and the same transfer $t^*(p_r, \epsilon)$ following a good environmental performance as if monitors were benevolent. Payments and rent are thus respectively given by (3) and (4).

Investigation probabilities. Turning now to the optimal investigation probabilities, we get:

Proposition 2 *The probabilities of investigation under integration are lower than with benevolent monitors when collusion is a concern:*

$$p_r^I \le p_r^* \text{ and } p_j^I = 0. \tag{12}$$

Considering the probability of ex ante investigation, several opposite effects are simultaneously at play. For a given stake of collusion between the merged entity and the firm, reducing this probability helps decreasing the expected wage $V(p_r, p_j)$ needed to ensure ex ante collusion-proofness. As with benevolent monitors, reducing this probability increases the incentive reward $t^*(p_r, \epsilon)$ and makes the incentive constraint (2) more demanding. However, with corruptible monitors, the stake of collusion $\pi_0 t^*(p_r, \epsilon)$ is increased, making the collusion-proofness constraint (9) more difficult to satisfy too. Since the elasticity of the reward with respect to the ex ante audit probability is smaller than one, the first of these effects always dominates, making the impact of p_r on the stake of collusion less important. Overall, this pushes towards less ex ante investigation.

4.3 Separation

Consider now the case where each round of investigation is run by a different monitor. The key difference with the case of integration is that, when deciding whether to collude or not with the ex ante monitor, the firm anticipates the issue of side-contracting with the ex post monitor. More precisely, we will assume that the ex ante monitor and the firm, at the time of striking their deals, anticipate that the ex post investigator will not collude in a collusion-proof equilibrium. Reciprocally, the ex post monitor anticipates that the ex ante monitor is not corrupted. These simple behavioral assumptions, where both agencies choose ex ante whether to collude or not, will help us to get clear results on the benefits of separation.

Ex post collusion. Absent any ex ante investigation, or when such investigation has failed, the same logic as under integration applies and the corresponding ex post collusion-proofness constraint is again given by (5).

Ex ante collusion. A side-contract for the ex ante monitor stipulates a priori some non-negative bribes $(\tau_r^0, \tau_r^1, \tau_r^2)$ respectively when an accident does not occur and when an accident does occur but ex post investigation either fails or succeeds.

Consider the last possibility. Through his own investigation, the ex post monitor not only observes the firm's effort but also the bribe τ_r^2 exchanged if any, detecting

thereby possible collusion between the ex ante monitor and the firm. Anticipating that the ex post monitor is uncorruptible in equilibrium, the firm and the ex ante monitor should strike their best collusive deal knowing that the ex post monitor is committed to report information on their misbehavior when detected. In particular, we assume that the regulator would lose all collusive benefits (may it be under the form of reputational stigma or financial penalties) following such a detection. From the firm's viewpoint, transferring bribe to that ex ante monitor in the event where the ex post monitor investigates is thus just a pure loss. It does not raise the ex ante monitor's expected benefit from colluding and only reduces the firm's expected payoff. Hence, the firm finds it optimal not to bribe the ex ante monitor in case an ex post investigation succeeds.

Therefore, the optimal side-contract (τ_r^0, τ_r^1) offered to the ex ante monitor when he observes $\sigma_r = 0$ and pretends that $\sigma_r = \emptyset$ must solve:

$$(\mathcal{P}^{S}): \max_{\tau_{r}^{0}, \tau_{r}^{1} \in [0,w]} w + \pi_{0}(t - \tau_{r}^{0}) - (1 - \pi_{0})(1 - p_{j}\epsilon)\tau_{r}^{1}$$

subject to $\pi_{0}k(\tau_{r}^{0}) + (1 - \pi_{0})(1 - p_{j}\epsilon)k(\tau_{r}^{1}) \geq V_{r}.$ (13)

Separation reduces the possibilities for smoothing bribes with respect to integration. Bribes with the ex ante monitor are now exchanged in two states of nature instead of three. This captures the intuition that, under separation, there are less points of contact and thus less collusion possibilities between the firm and its ex ante monitor.

Nevertheless, the dead-weight loss of side-contracting is still minimized with a flat bribe over contingencies where such a collusion is feasible:

$$\tau_r^0 = \tau_r^1 \equiv \tau_r^S = k^{-1} \left(\frac{V_r}{\alpha}\right)$$

where $\alpha = \pi_0 + (1 - \pi_0)(1 - p_j \epsilon)$ is the overall probability of no successful ex post investigation.

The firm's net benefit from ex ante collusion is thus $\pi_0 t - \alpha \tau_r^S$. The ex ante collusion-proofness constraint can thus be written as:

$$V_r \ge \alpha k \left(\frac{\pi_0 t}{\alpha}\right). \tag{14}$$

Response to the threat of capture. The overall cost of preventing capture is again obtained when the expected wage left to monitors is minimized, which implies that (5) and (14) are both binding:

$$V_r^S = \alpha k \left(\frac{\pi_0 t}{\alpha}\right) \text{ and } V_j^S = 0.$$
 (15)

Proceeding as before, the generalized incentive constraint that induces the firm to implement some care and not to collude with both the ex ante and ex post monitors is

similar to (10). This incentive constraint writes as

$$U \ge p_r \epsilon \max\left\{0; \pi_0 t - \alpha k^{-1} \left(\frac{V_r}{\alpha}\right)\right\} + (1 - p_r \epsilon) \left(\pi_0 t + (1 - \pi_0) p_j \epsilon \max\left\{0; -k^{-1}(V_j)\right\}\right).$$

It again boils down to the standard incentive constraint (4) when (5) and (14) are both binding. The optimal incentive transfer is thus unchanged and still given by (3). Hence, V_r^S becomes a function of (p_r, p_j) through the direct dependence of $t^*(p_r, \epsilon)$ on p_r and the dependence of α on p_j .

5 Welfare Comparison

Only the wages left for ex ante monitoring differ in both scenarios. For some fixed investigation probabilities (p_r, p_j) , the welfare difference between separation and integration is just the difference in these wages, which are respectively given by $V_r^I(p_r) = k (\pi_0 t^*(p_r, \epsilon))$ under integration, and $V_r^S(p_r, p_j) = \alpha k \left(\frac{\pi_0 t^*(p_r, \epsilon)}{\alpha}\right)$ under separation. We obtain:

$$\Delta W(p_r, p_j) = p_r \epsilon \left\{ V_r^I(p_r) - V_r^S(p_r, p_j) \right\} = p_r \epsilon \left\{ k \left(\pi_0 t^*(p_r, \epsilon) \right) - \alpha k \left(\frac{\pi_0 t^*(p_r, \epsilon)}{\alpha} \right) \right\}.$$

Since $\alpha < 1$, $t^*(p_r, \epsilon) > 0$ and $k(\cdot)$ is strictly concave, $\Delta W > 0$ and we can establish the main result of this paper:

Theorem 1 Under the threat of capture, and for any probabilities of investigation (p_r, p_j) , separation improves welfare:

$$\Delta W(p_r, p_j) > 0.$$

Transaction costs of side-contracting are greater under separation than under integration. The mere fact that the ex ante monitor and the firm cannot exchange bribes in all the states of nature makes it more difficult to collude ex ante.

The convexity of transaction costs is key for this result. Had transaction costs been linear, non-benevolent monitors would not care about smoothing bribes and separation would be equivalent to integration. Indeed, let us compare the "potential" bribe that the firm provides to the ex ante monitor under separation and integration. It is straightforward to see that:

$$\tau_r^I = \alpha \tau_r^S. \tag{16}$$

Therefore, the ex ante monitor under separation earns the same expected bribe than the ex ante monitor under integration. Indeed, in both cases, the firm wants to secure the liability rent $\pi_0 t^*(p_r, \epsilon)$. Comparing separation with integration, having higher bribes but less often is not costly when $k(\cdot)$ is linear.

However, two different effects that reinforce each other ensure that separation dominates when transaction costs are strictly convex. To understand those effects, remember that, although the same expected amount of bribes has to be distributed under integration and separation, there are fewer points of contact between the colluding partners under separation. Hence, the size of the bribe in each state of nature where collusion arises under separation increases significantly. This mechanically increases the deadweight loss of side-contracting, since it becomes at the margin harder to transfer such larger bribes. This is the first benefit of separation. However, a second benefit now appears since, compared with the case of integration, the distribution of those bribes exchanged with the ex ante monitor has also more variance (it is either 0 when ex post monitoring is successful or τ_r^S otherwise). This randomness increases the expected transaction costs because of their convexity.

Investigation probabilities. To give some preliminary insights on how the optimal probabilities of investigation under separation and integration can be compared, consider the simple case where k(.) is quadratic, namely, $k(\tau) = \kappa \tau - \frac{\mu}{2}\tau^2$ for some $\kappa \in [0, 1]$ and $\mu \ge 0$ such that $k(\cdot)$ is increasing on the relevant range. The welfare gain of separation can finally be written as:

$$\Delta W(p_r, p_j) = \frac{\mu \pi_0^2 (1 - \pi_0) (t^*(p_r, \epsilon))^2}{2(\pi_0 + (1 - \pi_0)(1 - p_j \epsilon))} p_r p_j \epsilon^2 > 0.$$
(17)

Those gains increase with the probability of an ex ante investigation since one can save on monitoring wages by splitting monitoring tasks under separation. In addition, when an ex post investigation is more likely, the transaction costs of side-contracting increase because the possibility of smoothing bribes under separation only occurs under contingencies which are relatively less likely. Bringing the ex post monitor in more often becomes thus more attractive. Denoting by (p_r^S, p_j^S) the optimal investigation probabilities under separation, we finally get:

Proposition 3 Assume that $k(\cdot)$ is quadratic. Under separation, there is more ex post investigation than under integration and more ex ante investigation if ϵ is small enough:

$$p_i^S > 0$$
 and $p_r^S > p_r^I \Leftrightarrow \Delta \pi > \pi_0 p_r^S \epsilon$.

From (17), we immediately deduce that the welfare gain from a marginal increase of the ex post investigation under separation is positive, i.e. $\frac{\partial \Delta W}{\partial p_j} > 0$. Instead, increasing p_r decreases the stake of collusion $\pi_0 t^*(p_r, \epsilon)$ by reducing the limited liability rent. However, increasing p_r also raises the probability of ex ante collusion, but it is less problematic under separation than under integration. The second of these effects dominates, and $\frac{\partial \Delta W}{\partial p_r} > 0$ when ϵ is small enough, since the elasticity of $t^*(p_r, \epsilon)$ with respect to p_r is then small.

A recurrent question concerning the use of regulators or judges is whether they are substitutes or complements. Our model sheds some light on this issue. The formula (17) shows under which conditions these two bodies are actually complements in improving welfare. Indeed, when ϵ is small enough, it becomes more valuable to use ex ante investigation under separation because the stake of collusion is reduced. At the same time, ex post intervention is more attractive. By raising at the margin the probability of bringing the judge in ex post, ex ante collusion with the regulator becomes

harder, making regulation more attractive.

When monitoring is not too efficient, either because the optimal ex ante audit probability is small or because the accuracy of the signal is weak, the condition $\Delta \pi > \pi_0 p_r^S \epsilon$ holds and ex ante and ex post interventions are complements.

It is striking that, even though the judge himself may be corruptible, an ex post investigation is now valuable, contrary to what happens in a collusion-free environment. There is an interesting division of tasks between public bodies: ex ante regulation helps relaxing the firm's incentive constraint whereas ex post judicial investigation helps relaxing the possible collusion between the firm and its regulator.

Finally, we close this section by considering more general formulations of our results.

Asymmetric signal accuracies. Let us investigate in more details the role played by the relative accuracy of the ex ante and the ex post monitoring. Denote by ϵ_r and ϵ_j the respective precisions of the ex ante and ex post signals available to the monitors. It is straightforward to check that the optimal incentive reward remains $t^*(p_r, \epsilon_r)$, which only depends on the accuracy of ex ante monitoring. Intuitively, ex post monitoring does not relax the firm's incentive constraint, as we showed earlier on, and the precision of the corresponding signal plays no role in determining the firm's reward and liability rent. Instead, notice that the overall probability of no successful investigation ex post $\alpha = \pi_0 + (1 - \pi_0)(1 - p_j \epsilon_j)$ depends only and decreases with the ex post accuracy ϵ_j . Still in the quadratic case, the welfare difference between separation and integration now becomes:

$$\Delta W(p_r, p_j) = \frac{\mu \pi_0^2 (1 - \pi_0) (t^*(p_r, \epsilon_r))^2}{2(\pi_0 + (1 - \pi_0)(1 - p_j \epsilon_j))} p_r p_j \epsilon_r \epsilon_j > 0.$$
(18)

Of course, this expression confirms that the ex ante and ex post rounds of monitoring remain complements over some range and that the welfare gain from separation increases with both degrees of accuracy. However, equation (18) shows also that this welfare gain is more sensitive to an increase in ϵ_j . Indeed, such a change decreases the conditional probability α of the states where bribes might be exchanged between the ex ante regulator and the firm, and thus increases the potential bribe τ^S . This allows to benefit from larger transaction costs of side-contracting at those high levels of bribes under separation.

General transaction costs. Formula (17) and (18) are obtained with a quadratic $k(\cdot)$. To understand the gains from separation beyond this simple functional form, it is useful to consider the limiting case where α is close to one, i.e. ϵ is close to zero. In that case, conditional on collusion having started with the ex ante monitor (admittedly, a rare event), the ex post round of investigation is unlikely to bring any information. Taylor expansions in the neighborhood of $\alpha = 1$ immediately give us:

$$\Delta W(p_r, p_j) = (1 - \alpha) \underbrace{\left(k(\pi_0 t^*(p_r, \epsilon)) - \pi_0 t^*(p_r, \epsilon)k'(\pi_0 t^*(p_r, \epsilon))\right)}_{\text{First-order effect}}$$

$$+\frac{(1-\alpha)^2}{2}\underbrace{(\pi_0 t^*(p_r,\epsilon))^2(-k''(\pi_0 t^*(p_r,\epsilon)))}_{\text{Second-order effect}}.$$

The first of these terms represents the first-order effect of reducing the overall probability of the states where bribes can be exchanged with the ex ante monitor by raising a bit the accuracy of the ex post signal. It captures the fact that the bribe possibly exchanged with the ex ante monitor being now higher under separation, transaction costs of sidecontracting are higher at those higher levels. The second term represents instead the second-order impact of reducing this probability. It is related to the greater variance of the distribution of bribes exchanged under separation. These two effects always go in the same direction. The first-order effect and the second-order effect are equal in the quadratic case, but they may differ beyond.

6 Discussion and Extensions

Testable implications. Our analysis provides a number of testable implications. Suppose that separating ex ante and ex post monitoring requires setting up a new agency with a fixed set-up cost K > 0. The formula (17) shows that $\Delta W(p_r, p_j)$ is more likely to be greater than K under several circumstances which can be related to the monitoring and production technologies available in the public and the private spheres. First, separation is attractive when p_r and p_j are both large enough, i.e. when the administrative costs of monitoring are small enough. We expect thus separation to be more easily adopted in mature sectors, where routinized administrative procedures may have lowered those marginal costs of monitoring. Transportation may be an example in order. Second, separation is also more attractive when $t^*(p_r, \epsilon)$ is high enough, i.e., when regulation calls for high powered incentives, maybe because the firm's marginal cost of effort is high or significant investments are needed to reduce the likelihood of an environmental accident. The nuclear industry is an attractive example in this respect.

Another important implication of our findings is that splitting ex ante and ex post investigations should increase the likelihood to rely on either of those arms and, by the same token, their administrative costs. The flip side of separation is thus a possible increase in regulatory budgets.

Robustness 1: Monitors' anticipatory behavior. We have assumed so far that the ex ante monitor and the firm, at the time of striking their deals, anticipate that the ex post investigator will not collude. Reciprocally, the ex post monitor also anticipates that the ex ante monitor is not corrupted. These simple behavioral assumptions, where both agencies can commit on their choice whether to collude or not, helped us to get clear results on the benefits of separation.

Let us now turn to the more intricate hypothesis where the ex ante monitor and the firm design their collusive deal anticipating its impact on the incentives to collude or not with the ex post monitor. The firm may then be able to bribe the ex ante monitor under all circumstances if it also bribes the ex post monitor and thereby buys his leniency. Of course, preventing such a behavior requires to increase the expost monitor's wage, as we show below.

Based on our previous findings, it is relatively straightforward to compute the firm's overall payoff when buying the expost monitor's services with a bribe $k^{-1}(V_i)$ if it also colludes with the ex ante monitor, giving the latter a flat bribe $k^{-1}(V_r)$ in all states of nature.³⁷ Since bribing the ex post monitor arises only with probability $1 - \alpha = (1 - \pi_0)p_i\epsilon_i$, the firm's payoff writes as:

$$w + \pi_0 t - k^{-1}(V_r) - (1 - \alpha)k^{-1}(V_j).$$

Instead, when not buying the ex post monitor's leniency, the firm can only exchange with the ex ante monitor a flat bribe $k^{-1}\left(\frac{V_r}{\alpha}\right)$ in the two remaining states of nature where this ex post monitor does not get information, i.e. with overall probability α . The firm gets thereby:

$$w + \pi_0 t - \alpha k^{-1} \left(\frac{V_r}{\alpha} \right).$$

Clearly, the firm does not find it attractive to collude with the ex post monitor when the second payoff dominates. This yields the following expression of the ex post collusion-proofness constraint:

$$V_j \ge k \left(\frac{1}{1 - \alpha} \left(\alpha k^{-1} \left(\frac{V_r}{\alpha} \right) - k^{-1} \left(V_r \right) \right) \right).$$
(19)

Together with the ex ante collusion-proofness constraint (14), this defines the set of wages ensuring that no collusion takes place either ex ante or ex post under separation.

Note that the strict convexity of $k^{-1}(\cdot)$, the fact that k(0) = 0 and $\alpha < 1$ altogether imply that the right-hand side of (19) is positive when $V_r > 0$. This condition is satisfied since (14) is binding. The ex post monitor must now receive a positive wage which is large enough to prevent a bribe exchange between the ex ante monitor and the firm in all states of nature. The benefits of lower payments to the ex ante monitor must be weighted against the cost of higher payments for the ex post monitor. The first effect dominates when $p_r \epsilon$ is large enough, in which case an expost investigation is unlikely, and leaving a wage to the ex post monitor is not so costly.

As an example, assume that transaction costs of side-contracting are quadratic, $k(\tau) = \tau - 1/2\tau^2$, and that bribes are small enough.³⁸ Then, $k^{-1}(v)$ can be approximated (up to terms of order more than 2) by $v + 1/2v^2$. In equilibrium, the ex ante monitor and the firm do not collude (while anticipating that the ex post monitor is not corrupted because (19) holds) if the ex ante collusion-proofness (14) is satisfied. When bribes are small, the ex post collusion-proofness constraint (19) can be approximated as: $V_j \ge V_r^2/(2\alpha)$. Simple manipulations then show that the following approximation

³⁷Here, we assume for simplicity that the firm has enough funds to bribe both monitors ex post, i.e., $w > k^{-1}(V_r) + k^{-1}(V_j)$ for the equilibrium values of wages. ³⁸This arises, for instance, when ψ is small so that $\pi_0 t^*(p_r, \epsilon)$ is also small.

holds also:

$$\Delta W(p_r, p_j) \approx \frac{\pi_0^2 (t^*(p_r, \epsilon))^2}{2\alpha} [(1 - \alpha)p_r - (1 - p_r \epsilon)(1 - \pi_1)p_j]\epsilon.$$

When p_r is large enough, $\Delta W(p_r, p_j) > 0$. In this case, separation dominates integration, but it now requires to leave a positive wage to the expost monitor.

Robustness 2: Monitors' wages. We have assumed implicitly throughout the exposition that monitors' wages do not depend on the firm's effort. This is without loss of generality when there is no collusion. This might seem a priori restrictive otherwise. Suppose indeed that the firm's effort, once observed, can be used to condition the monitor's compensation. With a wage dependent on the effort level, collusion can be prevented at no cost by offering a high wage \overline{V} to the monitor when he reveals that the firm has shirked. When the firm's individual incentive constraint is satisfied, this event does not occur along the equilibrium path and raising this wage has no cost for society (see Kessler, 2000). However, when evidence on effort can be manipulated by the monitor, things are different. Suppose that the signal $\sigma_i = e = 1$ (i = r, j) is only partially verifiable and can be manipulated into a report $\hat{\sigma}_i = e = 0$. The scheme above is no longer attractive. Indeed, inducing the monitor to reveal that the firm has actually complied with the standard requires to give him the wage \bar{V} . Now the wage \bar{V} is offered on the equilibrium path, which is socially costly. Hence, in settings where the auditor's information on the firm's effort is (partially) manipulable, the auditors' wages should not depend on the firm's effort.³⁹

7 Conclusion

This paper has stressed the benefits of splitting ex ante and ex post monitoring of environmentally risky ventures in a moral hazard environment. Having an independent ex post monitor intervening only upon an accident makes it more difficult for the firm to collude with the ex ante monitor whose control is more routinized. Regulatory capture is less of a concern under separation and this institutional choice improves social welfare.

Although our model generates some value for separation to improve the fight against capture, it is worth stressing other potential benefits from separation that could be added in a more complete model. First, separation may help to generate evidence because it allows to cross-check the monitors' announcements.⁴⁰ Second, duplication of expertise between ex ante and ex post monitors may facilitate specialization in gathering information on different dimensions of the firm's activities. The ex ante regulators are certainly more prone to gather technical information, whereas ex post judges

³⁹Another justification for ruling out compensations contingent on effort is that they may have a true cost if the firm shirks on effort with some probability on the equilibrium, either because it "trembles" a little bit or because the compliance cost is uncertain.

⁴⁰For a similar argument, see Laffont (2000).

would instead focus on testimonies by private parties. Investigating both the incentives for specialization and its consequences for institutional design would certainly be a valuable extension to our analysis.

From a theoretical viewpoint, our approach for modeling collusive behavior between the private sector and public bodies differs significantly from other studies of corruption in the regulation or law enforcement literature, which have instead stressed the equilibrium nature of corruption.⁴¹ In our paper, collusion is prevented by an adequate design of incentive rewards and comparison between institutions amounts to looking for the organizational form that minimizes the cost of preventing collusion. It would be interesting to investigate environments where collusion would be an equilibrium phenomenon. This could be done by introducing some non-observable heterogeneity among monitors, for instance, in terms of their willingness to collude or in terms of their psychological costs of being caught lying.⁴² How institutions change the equilibrium level of corruption is then an important topic that would be worth investigating.

References

- Baron, D. (1989), "Design of Regulatory Mechanisms and Institutions," *Handbook of Industrial Organization, Volume II*, R. Schmalensee and R. Willig eds., North Holland.
- Baron, D. and D. Besanko (1984), "Regulation, Asymmetric information and Auditing", *RAND Journal of Economics*, 15: 447-470.
- Baron, D. and R. Myerson (1982), "Regulating a Monopolist with Unknown Costs," *Econometrica*, 50: 911-930.
- Beck, U. (1992), *Risk Society, Towards a New Modernity*, London: Sage Publications.
- Becker, G. (1968), "Crime and Punishment: An Economic Approach", *Journal of Political Economy*, 76: 169-217.
- Becker, G. and G. Stigler (1974), "Law, Enforcement, Malfeasance and Compensation of Enforcers", *Journal of Legal Studies*, 3: 1-18.
- Boyer, M. and D. Porrini (2001), "Law versus Regulation: A Political Economy Model of Instruments Choice in Environmental Policy", in A. Heyes (ed.), *Law and Economics of the Environment*, Edward Elgar Publishing Ltd.
- Boyer, M. and D. Porrini (2004), "Modelling the Choice Between Regulation and Liability in Terms of Social Welfare", *Canadian Journal of Economics*, 37: 590-612.

⁴¹Among others, Becker and Stigler (1974), Mookherjee and P'Ng (1995), Garoupa (1997), Polinsky and Shavell (2001).

⁴²See Tirole (1992) and Kofman and Lawarée (1996b) for such models.

- Congleton, R. (1984), "Committees and Rent-Seeking Effort", *Journal of Public Economics*, 25: 197-209.
- Faure-Grimaud, A., J.-J. Laffont and D. Martimort (2002), "Risk Averse Supervisors and the Efficiency of Collusion", *Contributions in Theoretical Economics*. http://www.bepress.com/bejte/contributions/vol2/iss1/art5.
- Faure-Grimaud, A. and D. Martimort (2003), "Regulatory Inertia", *RAND Journal of Economics*, 34: 414-442.
- Garoupa, N. (1997), "The Theory of Optimal Law Enforcement", *Journal of Economic Surveys*, 11: 267-295.
- Hiriart, Y. and D. Martimort (2006), "The Benefits of Extended Liability", *Rand Journal of Economics*, 37: 562-582.
- Hiriart, Y., D. Martimort and J. Pouyet (2010), "Weak Enforcement of Environmental Policies: A Tale of Limited Commitment and Limited Fines," forthcoming *Annales d'Économie et de Statistique*.
- Hood, C., H. Rothstein and R. Baldwin (2001), *The Government of Risk*, Oxford University Press.
- Kessler, A. (2000), "On Monitoring and Collusion in Hierarchies," *Journal of Economic Theory*, 91: 280-292.
- Khalil, F. and J. Lawarée (2006), "Incentives for Corruptible Auditors in the Absence of Commitment," *Journal of Industrial Economics*, 54: 269-291.
- Kofman, F. and J. Lawarée (1993), "Collusion in Hierarchical Agency," *Econometrica*, 61: 629-656.
- Kofman, F. and J. Lawarée (1996a), "A Prisoner's Dilemma Model of Collusion Detterence," *Journal of Public Economics*, 59: 117-136.
- Kofman, F. and J. Lawarée (1996b), "On the Optimality of Allowing Collusion," *Journal of Public Economics*, 61: 384-407.
- Kolstad, C., T. Ulen and G. Johnson (1990), "Ex Post Liability for Harm versus Ex Ante Safety Regulation: Substitutes or Complements?" *American Economic Review*, 80: 888-901.
- Laffont, J.-J. (1990), "Analysis of Hidden-Gaming in a Three-Tier Hierarchy", *Journal of Law, Economics and Organization*, 6: 301-324.
- Laffont, J.-J. (2000), Incentives in Political Economy, Oxford University Press.
- Laffont, J.-J. and D. Martimort (1997), "Collusion under Asymmetric Information", *Econometrica*, 65: 875-911.
- Laffont, J.-J. and D. Martimort (1998), "Transaction Costs, Institution Design and the Separation of Powers", *European Economic Review*, 42: 673-684.
- Laffont, J.-J. and D. Martimort (1999), "Separation of Regulators Against Collusive Behavior", *RAND Journal of Economics*, 30: 232-262.

- Laffont, J.-J. and D. Martimort (2000), "Mechanism Design with Collusion and Correlation", *Econometrica*, 68: 309-342.
- Laffont, J.-J. and J. Tirole (1993), A Theory of Incentives in Procurement and Regulation, MIT Press.
- Lesourd, J.-B. and G. Schilizzi (2001), *The Environment in Corporate Management*, Edward Elgar.
- McLeod, B. (2003), "Optimal Contracting with Subjective Evaluation", American Economic Review, 93: 216-240.
- Mookherjee, D. and I. P'Ng (1992), "Monitoring vis--vis Investigation in Enforcement of Law", *American Economic Review*, 82: 556-565.
- Mookherjee, D. and I. P'Ng (1995), "Corruptible Law Enforcers: How Should They Be Compensated?", *The Economic Journal*, 105: 145-159.
- Niskanen, W. (1971), Bureaucracy and Representative Government, Aldine Press.
- Noll, R. (1989), "Economic Perspectives on the Politics of Regulation," *Handbook of Industrial Organization, Volume II*, R. Schmalensee and R. Willig eds., North Holland.
- Pitchford, R. (2001), "How Liable Should a Lender be? The Case of Judgment-Proof Firms and Environmental Risks: Reply," American Economic Review, 91: 739-745.
- Polinsky, M. and S. Shavell (2001), "Corruption and Optimal Law Enforcement", *Journal of Public Economics*, 81: 1-24.
- Power, M. (1997), The Audit Society, Oxford University Press.
- Shavell, S. (1984a), "A Model of the Optimal Use of Liability and Safety Regulation", *RAND Journal of Economics*, 15: 271-280.
- Shavell, S. (1984b), "Liability for Harm versus Regulation of Safety", *Journal* of Legal Studies, 13: 357-374.
- Shleifer, A. and R. Vishny (1993), "Corruption", *Quarterly Journal of Economics*, 108: 599-617.
- Strausz, R. (1997), "Collusion and Renegotiation in a Principal-Supervisor-Agent Relationship," *Scandinavian Journal of Economics*, 99: 497-518.
- Tirole, J. (1986), "Hierarchies and Bureaucracies", *Journal of Law, Economics and Organization*, 2: 181-214.
- Tirole, J. (1992), "Collusion and the Theory of Organizations", in Advances in Economic Theory: Proceedings of the Sixth World Congress of the Econometric Society, ed. J.-J. Laffont, Cambridge: Cambridge University Press.
- Vafai, K. (2002), "Preventing Abuse of Authorities in Hierarchies," *International Journal of Industrial Organization*, 20: 1143-1166.
- Wittman, D. (1977), "Prior Regulation versus Post Liability: The Choice between Input and Output Monitoring", *Journal of Legal Studies*, 6: 193-212.

Appendix

Proof of Proposition 1. We first compute the agency cost necessary to implement the high level of safety care given the investigation probabilities. Second, we determine optimal probabilities taking into account agency and administrative costs.

For fixed probabilities of inspection (p_r, p_j) the high level of care is implemented at a social cost $C^*(p_r, p_j)$ such that:

$$\mathcal{C}^*(p_r, p_j) \equiv \min_{\{U, V_r, V_j\}} (1 - \gamma)U + V(p_r, p_j) + C(p_r) + C(p_j) \text{ subject to (1) and (4)}.$$

When monitors are benevolent and report truthfully any information they may gather on the firm's effort, wages are optimally set at zero ((1) is binding) so that $V(p_r, p_j) = 0$. The optimal probabilities of investigation (p_r^*, p_j^*) are given by:

$$(p_r^*, p_j^*) = \arg \min_{\{p_r, p_j\}} C^*(p_r, p_j) = (1 - \gamma)U(p_r) + C(p_r) + C(p_j).$$

First-order necessary (and sufficient given strict convexity of the objective function) conditions give us:

$$C'(p_r^*) = \frac{(1-\gamma)\pi_1\pi_0\psi\epsilon}{(\Delta\pi + \pi_0 p_r^*\epsilon)^2} > 0 \text{ and } C'(p_j^*) = 0.$$
(20)

The expression for $t^* = t^*(p_r^*, \epsilon)$ is given by (3) with the optimal probability p_r^* above.

Proof of Proposition 2. Under integration, and for some fixed probabilities of investigation, the optimal regulatory charter solves:

$$\mathcal{C}^{I}(p_{r}, p_{j}) \equiv \min_{\{t, V_{r}, V_{j}\}} (1 - \gamma)U + V(p_{r}, p_{j}) + C(p_{r}) + C(p_{j}) \text{ subject to (5), (9) and (10)}.$$

This immediately yields the expressions of wages for monitoring in (11). Taking into account those wages and the expression of the firm's liability rent, the optimal investigation probabilities solve:

$$\left(p_r^I, p_j^I\right) = \arg\min_{\{p_r, p_j\}} \mathcal{C}^I(p_r, p_j) \equiv \mathcal{C}^*(p_r, p_j) + p_r \epsilon k \left(\pi_0 t^*(p_r, \epsilon)\right).$$

The derivatives of this objective function w.r.t p_r and p_j are respectively given by:

$$\frac{\partial \mathcal{C}^{I}(p_{r},p_{j})}{\partial p_{r}} = \frac{\partial \mathcal{C}^{*}(p_{r},p_{j})}{\partial p_{r}} + \epsilon \left(k \left(\pi_{0} t^{*}(p_{r},\epsilon) \right) + \pi_{0} p_{r} \frac{\partial t^{*}(p_{r},\epsilon)}{\partial p_{r}} k' \left(\pi_{0} t^{*}(p_{r},\epsilon) \right) \right), \quad (21)$$

$$\frac{\partial \mathcal{C}^{I}(p_{r}, p_{j})}{\partial p_{j}} = \frac{\partial \mathcal{C}^{*}(p_{r}, p_{j})}{\partial p_{j}} = C'(p_{j}).$$
(22)

Note that

$$\eta_r \equiv -\frac{p_r}{t^*(p_r,\epsilon)} \frac{\partial t^*(p_r,\epsilon)}{\partial p_r} = \frac{\pi_0 p_r \epsilon}{\Delta \pi + \pi_0 p_r \epsilon} < 1.$$

Therefore, we get

$$k(\pi_{0}t^{*}(p_{r},\epsilon)) + \pi_{0}p_{r}\frac{\partial t^{*}(p_{r},\epsilon)}{\partial p_{r}}k'(\pi_{0}t^{*}(p_{r},\epsilon))$$

= $k(\pi_{0}t^{*}(p_{r},\epsilon)) - \eta_{r}\pi_{0}t^{*}(p_{r},\epsilon)k'(\pi_{0}t^{*}(p_{r},\epsilon))$
> $k(\pi_{0}t^{*}(p_{r},\epsilon)) - \pi_{0}t^{*}(p_{r},\epsilon)k'(\pi_{0}t^{*}(p_{r},\epsilon)) > 0,$

where the last inequality follows from the concavity of $k(\cdot)$. This implies $\frac{\partial \mathcal{C}^{I}}{\partial p_{r}}(p_{r}, p_{j}) > \frac{\partial \mathcal{C}^{*}}{\partial p_{r}}(p_{r}, p_{j})$ and thus $p_{r}^{I} < p_{r}^{*}$. Also, we immediately get $p_{j}^{I} = 0$.

Proof of Proposition 3. Under separation, and for some fixed probabilities of investigation, the optimal regulatory charter solves:

$$\mathcal{C}^{S}(p_{r}, p_{j}) \equiv \min_{\{t, V_{r}, V_{j}\}} (1 - \gamma)U + V(p_{r}, p_{j}) + C(p_{r}) + C(p_{j}) \text{ subject to (5), (10) and (14).}$$

This immediately yields the expressions of wages for monitoring in (15). Taking into account those wages and the expression of the firm's liability rent, the optimal investigation probabilities solve:

$$(p_r^S, p_j^S) = \arg\min_{\{p_r, p_j\}} \mathcal{C}^S(p_r, p_j) \equiv \mathcal{C}^I(p_r, p_j) + p_r \epsilon \left[\alpha k \left(\frac{\pi_0 t^*(p_r, \epsilon)}{\alpha} \right) - k(\pi_0 t^*(p_r, \epsilon)) \right]$$
$$\equiv \mathcal{C}^I(p_r, p_j) - \frac{(1 - \alpha)\mu(\pi_0 t^*(p_r, \epsilon))^2}{2\alpha} p_r \epsilon.$$

From this, we compute:

$$\frac{\partial \mathcal{C}^S(p_r, p_j)}{\partial p_r} = \frac{\partial \mathcal{C}^I(p_r, p_j)}{\partial p_r} - \frac{\mu \pi_0^2 (1 - \pi_0) \psi^2 (\Delta \pi - \pi_0 p_r \epsilon)}{2\alpha (\Delta \pi + \pi_0 p_r \epsilon)^3} p_j \epsilon^2$$

and thus $p_r^S > p_r^I \Leftrightarrow \Delta \pi > \pi_0 p_r^S \epsilon$. Also, we get:

$$\frac{\partial \mathcal{C}^S(p_r, p_j)}{\partial p_j} = C'(p_j) - \frac{\mu \pi_0^2 (1 - \pi_0) \psi^2}{2\alpha^2 (\Delta \pi + \pi_0 p_r \epsilon)^2} p_r \epsilon^2$$

This yields also $p_j^S > 0$.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/getpage.aspx?id=73&sez=Publications&padre=20&tab=1 http://papers.ssrn.com/sol3/JELJOUR_Results.cfm?form_name=journalbrowse&journal_id=266659

http://ideas.repec.org/s/fem/femwpa.html http://www.econis.eu/LNG=EN/FAM?PPN=505954494

http://ageconsearch.umn.edu/handle/35978

http://www.bepress.com/feem/

NOTE DI LAVORO PUBLISHED IN 2010

GC	1.2010	Cristina Cattaneo: <u>Migrants' International Transfers and Educational Expenditure: Empirical Evidence from</u>
		Albania
SD	2.2010	Fabio Antoniou, Panos Hatzipanayotou and Phoebe Koundouri: <u>Tradable Permits vs Ecological Dumping</u>
SD	3.2010	Fabio Antoniou, Panos Hatzipanayotou and Phoebe Koundouri: <u>Second Best Environmental Policies under</u>
SD	4 2010	Carlo Carraro, Enrica De Cian and Lea Nicita: Modeling Biased Technical Change Implications for Climate
50	4.2010	Policy
IM	5.2010	Luca Di Corato: Profit Sharing under the threat of Nationalization
SD	6.2010	Masako Ikefuji, Jun-ichi Itaya and Makoto Okamura: Optimal Emission Tax with Endogenous Location
		Choice of Duopolistic Firms
SD	7.2010	Michela Catenacci and Carlo Giupponi: Potentials and Limits of Bayesian Networks to Deal with
		Uncertainty in the Assessment of Climate Change Adaptation Policies
GC	8.2010	Paul Sarfo-Mensah and William Oduro: Changes in Beliefs and Perceptions about the Natural Environment
		<u>in the Forest-Savanna Transitional Zone of Ghana: The Influence of Religion</u>
IM	9.2010	Andrea Boitani, Marcella Nicolini and Carlo Scarpa: Do Competition and Ownership Matter? Evidence
		from Local Public Transport in Europe
SD	10.2010	Helen Ding and Paulo A.L.D. Nunes and Sonja Teelucksingh: <u>European Forests and Carbon Sequestration</u>
		Services : An Economic Assessment of Climate Change Impacts
GC	11.2010	Enrico Bertacchini, Walter Santagata and Giovanni Signorello: Loving Cultural Heritage Private Individual
	10.0010	Giving and Prosocial Behavior
SD	12.2010	Antoine Dechezleprêtre, Matthieu Glachant and Yann Ménière: <u>What Drives the International Transfer of</u>
C D	12 2010	Climate Change Mitigation Technologies? Empirical Evidence from Patent Data
SD	13.2010	Andrea Bastianin, Alice Favero and Emanuele Massetti: <u>Investments and Financial Flows Induced by</u>
CD	14 2010	Climate Mitigation Policies
SD	14.2010	Rever Genagn: <u>100 Much Oli</u>
	16.2010	Chiara Fumagaili and Massimo Motta: <u>A Simple Theory of Predation</u>
uc	10.2010	Rinaldo Drau, Adriana Di Liberto and Francesco Pigliaru: <u>Tourism and Development: A Recent</u>
SD	17 2010	<u>Prenomenon Built on Oid (Institutional) Roots:</u>
30	17.2010	MOSE Barriers on Venice Port Activities
SD	18 2010	ZhongXiang Zhang: Climate Change Meets Trade in Promoting Green Growth: Potential Conflicts and
50	10.2010	Svnergies
SD	19.2010	Elisa Lanzi and Ian Sue Wing: Capital Malleability and the Macroeconomic Costs of Climate Policy
IM	20.2010	Alberto Petrucci: Second-Best Optimal Taxation of Oil and Capital in a Small Open Economy
SD	21.2010	Enrica De Cian and Alice Favero: Fairness, Credibility and Effectiveness in the Copenhagen Accord: An
		Economic Assessment
SD	22.2010	Francesco Bosello: Adaptation, Mitigation and "Green" R&D to Combat Global Climate Change. Insights
		From an Empirical Integrated Assessment Exercise
IM	23.2010	Jean Tirole and Roland Bénabou: Individual and Corporate Social Responsibility
IM	24.2010	Cesare Dosi and Michele Moretto: Licences, "Use or Lose" Provisions and the Time of Investment
GC	25.2010	Andrés Rodríguez-Pose and Vassilis Tselios (lxxxvi): <u>Returns to Migration, Education, and Externalities in</u>
		the European Union
GC	26.2010	Klaus Desmet and Esteban Rossi-Hansberg (lxxxvi): <u>Spatial Development</u>
SD	27.2010	Massimiliano Mazzanti, Anna Montini and Francesco Nicolli: Waste Generation and Landfill Diversion
		Dynamics: Decentralised Management and Spatial Effects
SD	28.2010	Lucia Ceccato, Valentina Giannini and Carlo Gipponi: <u>A Participatory Approach to Assess the Effectiveness</u>
		of Responses to Cope with Flood Risk
SD	29.2010	Valentina Bosetti and David G. Victor: <u>Politics and Economics of Second-Best Regulation of Greenhouse</u>
		Gases: The Importance of Regulatory Credibility
IM	30.2010	Francesca Cornelli, Zbigniew Kominek and Alexander Ljungqvist: <u>Monitoring Managers: Does it Matter?</u>
GC	31.2010	Francesco D'Amuri and Juri Marcucci: "Google it!" Forecasting the US Unemployment Rate with a Google
50	22 2010	JOD Search Index
20	32.2010	Francesco Doseilo, Cario Carraro and Enrica De Cian: <u>Climate Policy and the Optimal Balance between</u>
		<u>willigation, Adaptation and Unavoided Damage</u>

SD	33.2010	Enrica De Cian and Massimo Tavoni: <u>The Role of International Carbon Offsets in a Second-best Climate</u>
SD	34.2010	ZhongXiang Zhang: The U.S. Proposed Carbon Tariffs, WTO Scrutiny and China's Responses
IM	35.2010	Vincenzo Denicolò and Piercarlo Zanchettin: Leadership Cycles
SD	36.2010	Stéphanie Monjon and Philippe Quirion: <u>How to Design a Border Adjustment for the European Union</u> Emissions Trading System?
SD	37.2010	Meriem Hamdi-Cherif, Céline Guivarch and Philippe Quirion: <u>Sectoral Targets for Developing Countries:</u> Combining "Common but Differentiated Responsibilities" with "Meaningful participation"
IM	38.2010	G. Andrew Karolyi and Rose C. Liao: <u>What is Different about Government-Controlled Acquirers in Cross</u> - Border Acquisitions?
GC	39,2010	Kietil Biorvatn and Alireza Naghavi: Rent Seekers in Rentier States: When Greed Brings Peace
GC	40.2010	Andrea Mantovani and Alireza Naghavi: Parallel Imports and Innovation in an Emerging Economy
SD	41.2010	Luke Brander, Andrea Ghermandi, Onno Kuik, Anil Markandya, Paulo A.L.D. Nunes, Marije Schaafsma and Alfred Wagtendonk: <u>Scaling up Ecosystem Services Values: Methodology, Applicability and a Case</u>
SD	42.2010	Valentina Bosetti, Carlo Carraro, Romain Duval and Massimo Tavoni: <u>What Should We Expect from</u> <u>Innovation? A Model-Based Assessment of the Environmental and Mitigation Cost Implications of Climate-</u>
		Related R&D
SD	43.2010	Frank Vöhringer, Alain Haurie, Dabo Guan, Maryse Labriet, Richard Loulou, Valentina Bosetti, Pryadarshi R. Shukla and Philippe Thalmann: <u>Reinforcing the EU Dialogue with Developing Countries on Climate</u>
GC	44.2010	Change Mitigation Angelo Antoci, Pier Luigi Sacco and Mauro Sodini: <u>Public Security vs. Private Self-Protection: Optimal</u>
IM	45 2010	<u>I axation and the Social Dynamics of Fear</u> Luca Enriques: European Takeover Law, The Case for a Neutral Approach
SD	46.2010	Maureen L. Cropper, Yi Jiang, Anna Alberini and Patrick Baur: Getting Cars Off the Road: The Cost-
02		Effectiveness of an Episodic Pollution Control Program
IM	47.2010	Thomas Hellman and Enrico Perotti: The Circulation of Ideas in Firms and Markets
IM	48.2010	James Dow and Enrico Perotti: <u>Resistance to Change</u>
SD	49.2010	Jaromir Kovarik, Friederike Mengel and José Gabriel Romero: (Anti-) Coordination in Networks
SD	50.2010	Helen Ding, Silvia Silvestri, Aline Chiabai and Paulo A.L.D. Nunes: <u>A Hybrid Approach to the Valuation of</u> <u>Climate Change Effects on Ecosystem Services: Evidence from the European Forests</u>
GC	51.2010	Pauline Grosjean (lxxxvii): <u>A History of Violence: Testing the 'Culture of Honor' in the US South</u>
GC	52.2010	Paolo Buonanno and Matteo M. Galizzi (Ixxxvii): <u>Advocatus, et non latro? Testing the Supplier-Induced-Demand Hypothesis for Italian Courts of Justice</u>
GC	53.2010	Gilat Levy and Ronny Razin (Ixxxvii): <u>Religious Organizations</u>
GC	54.2010	Matteo Cervellati and Paolo Vanin (lxxxvii): <u>"Thou shalt not covet": Prohibitions, Temptation and Moral Values</u>
GC	55.2010	Sebastian Galiani, Martín A. Rossi and Ernesto Schargrodsky (lxxxvii): <u>Conscription and Crime: Evidence</u> from the Argentine Draft Lottery
GC	56.2010	Alberto Alesina, Yann Algan, Pierre Cahuc and Paola Giuliano (Ixxxvii): <u>Family Values and the Regulation of</u> <u>Labor</u>
GC	57.2010	Raquel Fernández (lxxxvii): <u>Women's Rights and Development</u>
GC	58.2010	Tommaso Nannicini, Andrea Stella, Guido Tabellini, Ugo Troiano (Ixxxvii): <u>Social Capital and Political</u> <u>Accountability</u>
GC	59.2010	Eleonora Patacchini and Yves Zenou (Ixxxvii): Juvenile Delinquency and Conformism
GC	60.2010	Gani Aldashev, Imane Chaara, Jean-Philippe Platteau and Zaki Wahhaj (Ixxxvii): <u>Using the Law to Change the Custom</u>
GC	61.2010	Jeffrey Butler, Paola Giuliano and Luigi Guiso (Ixxxvii): <u>The Right Amount of Trust</u>
SD	62.2010	Valentina Bosetti, Carlo Carraio and Massimo Tavoni: <u>Alternative Paths toward a Low Carbon World</u>
SD	63.2010	Kelly C. de Bruin, Rob B. Dellink and Richard S.J. Tol: <u>International Cooperation on Climate Change</u> <u>Adaptation from an Economic Perspective</u>
IM	64.2010	Andrea Bigano, Ramon Arigoni Ortiz, Anil Markandya, Emanuela Menichetti and Roberta Pierfederici: <u>The</u> <u>Linkages between Energy Efficiency and Security of Energy Supply in Europe</u>
SD	65.2010	Anil Markandya and Wan-Jung Chou: <u>Eastern Europe and the former Soviet Union since the fall of the</u> Berlin Wall: <u>Review of the Changes in the Environment and Natural Resources</u>
SD	66.2010	Anna Alberini and Milan Ščasný: <u>Context and the VSL: Evidence from a Stated Preference Study in Italy and the Czech Republic</u>
SD	67.2010	Francesco Bosello, Ramiro Parrado and Renato Rosa: <u>The Economic and Environmental Effects of an EU</u> Ban on Illegal Logging Imports. Insights from a CGE Assessment
IM	68.2010	Alessandro Fedele, Paolo M. Panteghini and Sergio Vergalli: <u>Optimal Investment and Financial Strategies</u> under Tax Rate Uncertainty
IM	69.2010	Carlo Cambini, Laura Rondi: <u>Regulatory Independence and Political Interference</u> : <u>Evidence from EU Mixed-</u> <u>Ownership Utilities' Investment and Debt</u>
SD	70.2010	Xavier Pautrel: Environmental Policy, Education and Growth with Finite Lifetime: the Role of Abatement Technology
SD	71.2010	Antoine Leblois and Philippe Quirion: <u>Agricultural Insurances Based on Meteorological Indices:</u> <u>Realizations, Methods and Research Agenda</u>
IM IM	72.2010 73.2010	Bin Dong and Benno Torgler: <u>The Causes of Corruption: Evidence from China</u> Bin Dong and Benno Torgler: <u>The Consequences of Corruption: Evidence from China</u>

IM	74 2010	Fereveloup Verdineiad and Vasaman Gorij: The Oil-Based Economies International Research Project. The
IIVI	74.2010	Case of Iran
66	75 2010	<u>Case of Irall.</u>
uc	73.2010	Scenes Michaelopoulos, Amitza Ragnavi and Giovanni Prarolo (IXXXII). <u>Trade and Geography III die</u>
SD.	76 2010	Zhone Vienz Zhenzi China in the Transition to a Low Carbon Economy
30	70.2010	Zhong Zhang: <u>Chima in the transition to a Low-Carbon Economy</u>
SD	//.2010	Valentina latolla, Massimiliano Mazzanti and Francesco Nicolli: <u>Are You SURE You Want to Waste Policy</u>
		Chances? Waste Generation, Landfill Diversion and Environmental Policy Effectiveness in the EU15
IM	78.2010	Jean Tirole: <u>Illiquidity and all its Friends</u>
SD	79.2010	Michael Finus and Pedro Pintassilgo: International Environmental Agreements under Uncertainty: Does
		the Veil of Uncertainty Help?
SD	80.2010	Robert W. Hahn and Robert N. Stavins: The Effect of Allowance Allocations on Cap-and-Trade System
		Performance
SD	81 2010	Francisco Alpizar Fredrik Carlsson and Maria Naranio (Ixxxviji): The Effect of Risk Ambiguity and
50	01.2010	Coordination on Farmers' Adaptation to Climate Change: A Framed Field Experiment
۲D	82 2010	Coordination of markets Adaptation to cancer angle. A market field Experiment
30	82.2010	Shardu Agrawaa ahu Maelis Carraro (ixxxviii). Assessing the Role of Micromatice in Fostering Adaptation
CD	02 2010	to chimate Change
SD	83.2010	Wolfgang Lutz (IxxxvIII): Improving Education as Key to Enhancing Adaptive Capacity in Developing
		Countries
SD	84.2010	Rasmus Heltberg, Habiba Gitay and Radhika Prabhu (lxxxviii): <u>Community-based Adaptation: Lessons</u>
		from the Development Marketplace 2009 on Adaptation to Climate Change
SD	85.2010	Anna Alberini, Christoph M. Rheinberger, Andrea Leiter, Charles A. McCormick and Andrew Mizrahi:
		What is the Value of Hazardous Weather Forecasts? Evidence from a Survey of Backcountry Skiers
SD	86.2010	Anna Alberini, Milan Ščasný, Dennis Guignet and Stefania Tonin: The Benefits of Contaminated Site
02	00.2010	Cleanup Revisited: The Case of Naples and Caserta Italy
CC	97 2010	David Sorte Mansah William Odwar Erediak Antah Eredua and Stanhan Amisah. Traditional
uc	87.2010	Paul Sario-Mensan, William Oduro, Fredrick Anton Fredua and Stephen Antisan. Traditional
		Representations of the Natural Environment and Biodiversity Conservation: Sacred Groves in Ghana
IM	88.2010	Gian Luca Clementi, Thomas Cooley and Sonia Di Giannatale: <u>A Theory of Firm Decline</u>
IM	89.2010	Gian Luca Clementi and Thomas Cooley: <u>Executive Compensation: Facts</u>
GC	90.2010	Fabio Sabatini: Job Instability and Family Planning: Insights from the Italian Puzzle
SD	91.2010	ZhongXiang Zhang: Copenhagen and Beyond: Reflections on China's Stance and Responses
SD	92.2010	ZhongXiang Zhang: Assessing China's Energy Conservation and Carbon Intensity: How Will the Future
		Differ from the Past?
SD	93,2010	Daron Acemoglu, Philippe Aghion, Leonardo Bursztyn and David Hemous: The Environment and Directed
02	2012010	Technical Change
SD	94 2010	Valaria Costantiai and Massimiliano Mazzanti. On the Green Side of Trade Competitiveness?
30	94.2010	Valeria Costantini and massiminatio mazzanti. On the Green side of trade competitiveness:
15.4	05 2010	Environmental Policies and innovation in the EU
IM	95.2010	Vittoria Cerasi, Barbara Chizzolini and Marc Ivaldi: <u>The Impact of Mergers on the Degree of Competition</u>
		in the Banking Industry
SD	96.2010	Emanuele Massetti and Lea Nicita: The Optimal Climate Policy Portfolio when Knowledge Spills Across
		Sectors
SD	97.2010	Sheila M. Olmstead and Robert N. Stavins: Three Key Elements of Post-2012 International Climate Policy
		Architecture
SD	98.2010	Lawrence H. Goulder and Robert N. Stavins: Interactions between State and Federal Climate Change
		Policies
IM	99 2010	Philippe Aghion John Van Reenen and Luigi Zingales: Innovation and Institutional Ownership
<u> </u>	100 2010	Angele Antoni Eshio Selatini and Auge Sedini. The Selavia Sundrama Seciel Control in a Crawing
uc	100.2010	Angelo Antoci, Fabio Sabatim and Mauro Sodim: <u>The Solaria Syndrome: Social Capital in a Growing</u>
60	101 0010	<u>Hyper-tecnnological Economy</u>
SD	101.2010	Georgios Kossioris, Michael Plexousakis, Anastasios Xepapadeas and Aart de Zeeuw: On the Optimal
		Taxation of Common-Pool Resources
SD	102.2010	ZhongXiang Zhang: Liberalizing Climate-Friendly Goods and Technologies in the WTO: Product Coverage,
		Modalities, Challenges and the Way Forward
SD	103.2010	Gérard Mondello: Risky Activities and Strict Liability Rules: Delegating Safety
GC	104.2010	João Ramos and Benno Torgler: Are Academics Messy? Testing the Broken Windows Theory with a Field
		Experiment in the Work Environment
INA	105 2010	Mauria Cinechai Erangesca Severini Claudio Socci and Posita Pretaroli: The Economic Impact of the
IIVI	103.2010	Construction Made through the Mars Multiplice Assessed
CD	100 2010	Green Certificate Market through the Macro Multiplier Approach
SD	106.2010	Joele Noally: improving the Energy-Efficiency of Buildings: The Impact of Environmental Policy on
		lechnological Innovation
SD	107.2010	Francesca Sanna-Randaccio and Roberta Sestini: <u>The Impact of Unilateral Climate Policy with Endogenous</u>
		Plant Location and Market Size Asymmetry
SD	108.2010	Valeria Costantini, Massimiliano Mozzanti and Anna Montini: Environmental Performance and Regional
		Innovation Spillovers
IM	109.2010	Elena Costantino, Maria Paola Marchello and Cecilia Mezzano: Social Responsibility as a Driver for Local
		Sustainable Development
GC	110 2010	Marco Percoco: Path Dependence Institutions and the Density of Economic Activities: Evidence from
30	110.2010	Italian Citize
SD.	111 0010	Italian Critics
20	111.2010	Sonja S. reelucksingn and Paulo A.L.D. Nunes: <u>Biodiversity valuation in Developing Countries: A Focus</u>
CD	110 0010	on small Island Developing States (SIDS)
50	112.2010	Znongxiang Znang: In What Format and under What Timetrame Would China Take on Climate
		Commitments? A Roadmap to 2050

SD	113.2010	Emanuele Massetti and Fabio Sferra: <u>A Numerical Analysis of Optimal Extraction and Trade of Oil under</u> <u>Climate Policy</u>
IM	114.2010	Nicola Gennaioli, Andrei Shleifer and Robert Vishny: <u>A Numerical Analysis of Optimal Extraction and</u> Trade of Oil under Climate Policy
GC	115.2010	Romano Piras: <u>Internal Migration Across Italian regions: Macroeconomic Determinants and</u> Accommodating Potential for a Dualistic Economy
SD	116.2010	Messan Agbaglah and Lars Ehlers (Ixxxix): Overlapping Coalitions, Bargaining and Networks
SD	117.2010	Pascal Billand, Christophe Bravard, Subhadip Chakrabarti and Sudipta Sarangi (lxxxix): <u>Spying in Multi-</u> market Oligopolies
SD	118.2010	Roman Chuhay (Ixxxix): <u>Marketing via Friends: Strategic Diffusion of Information in Social Networks with</u> Homophily
SD	119.2010	Françoise Forges and Ram Orzach (Ixxxix): <u>Core-stable Rings in Second Price Auctions with Common</u> Values
SD	120.2010	Markus Kinateder (lxxxix): The Repeated Prisoner's Dilemma in a Network
SD	121.2010	Alexey Kushnir (Ixxxix): Harmful Signaling in Matching Markets
SD	122.2010	Emiliya Lazarova and Dinko Dimitrov (Ixxxix): <u>Status-Seeking in Hedonic Games with Heterogeneous</u> <u>Players</u>
SD	123.2010	Maria Montero (lxxxix): <u>The Paradox of New Members in the EU Council of Ministers: A Non-cooperative</u> <u>Bargaining Analysis</u>
SD	124.2010	Leonardo Boncinelli and Paolo Pin (lxxxx): <u>Stochastic Stability in the Best Shot Game</u>
SD	125.2010	Nicolas Quérou (lxxxix): <u>Group Bargaining and Conflict</u>
SD	126.2010	Emily Tanimura (lxxxix): <u>Diffusion of Innovations on Community Based Small Worlds: the Role of</u> <u>Correlation between Social Spheres</u>
SD	127.2010	Alessandro Tavoni, Maja Schlüter and Simon Levin (Ixxxix): <u>The Survival of the Conformist: Social Pressure</u> <u>and Renewable Resource Management</u>
SD	128.2010	Norma Olaizola and Federico Valenciano (Ixxxix): Information, Stability and Dynamics in Networks under Institutional Constraints
GC	129.2010	Darwin Cortés, Guido Friebel and Darío Maldonado (Ixxxvii): <u>Crime and Education in a Model of</u> <u>Information Transmission</u>
IM	130.2010	Rosella Levaggi, Michele Moretto and Paolo Pertile: <u>Static and Dynamic Efficiency of Irreversible Health</u> <u>Care Investments under Alternative Payment Rules</u>
SD	131.2010	Robert N. Stavins: The Problem of the Commons: Still Unsettled after 100 Years
SD	132.2010	Louis-Gaëtan Giraudet and Dominique Finon: <u>On the Road to a Unified Market for Energy Efficiency: The</u> <u>Contribution of White Certificates Schemes</u>
SD	133.2010	Melina Barrio and Maria Loureiro: <u>The Impact of Protest Responses in Choice Experiments</u>
IM	134.2010	Vincenzo Denicolò and Christine Halmenschlager: <u>Optimal Patentability Requirements with Fragmented</u> <u>Property Rights</u>
GC	135.2010	Angelo Antoci, Paolo Russu and Elisa Ticci: <u>Local Communities in front of Big External Investors: An</u> <u>Opportunity or a Risk?</u>
SD	136.2010	Carlo Carraro and Emanuele Massetti: <u>Beyond Copenhagen: A Realistic Climate Policy in a Fragmented</u> <u>World</u>
SD	137.2010	Valentin Przyluski and Stéphane Hallegatte: <u>Climate Change Adaptation</u> , <u>Development</u> , <u>and International</u> <u>Financial Support</u> : <u>Lessons from EU Pre-Accession and Solidarity Funds</u>
SD	138.2010	Ruslana Rachel Palatnik and Paulo A.L.D. Nunes: <u>Valuation of Linkages between Climate Change</u> , <u>Biodiversity and Productivity of European Agro-Ecosystems</u>
SD	139.2010	Anna Alberini and Milan Ščasný: <u>Does the Cause of Death Matter? The Effect of Dread, Controllability,</u> <u>Exposure and Latency on the Vsl</u>
IM	140.2010	Gordon L. Clark and Ashby H. B. Monk: <u>Sovereign Wealth Funds: Form and Function in the 21st Century</u>
SD	141,2010	Simone Borghesi: <u>The European Emission Trading Scheme and Renewable Energy Policies: Credible</u> <u>Targets for Incredible Results?</u>
SD	142.2010	Francesco Bosello and Fabio Eboli: <u>REDD in the Carbon Market: A General Equilibrium Analysis</u>
SD	143.2010	Irene Valsecchi: <u>Repeated Cheap-Talk Games of Common Interest between a Decision-Maker and an</u> <u>Expert of Unknown Statistical Bias</u>
IM	144.2010	Yolande Hiriart, David Martimort and Jerome Pouyet: <u>The Public Management of Risk: Separating Ex Ante</u> and Ex Post Monitors

(lxxxvi) This paper was presented at the Conference on "Urban and Regional Economics" organised by the Centre for Economic Policy Research (CEPR) and FEEM, held in Milan on 12-13 October 2009.

(lxxxvii) This paper was presented at the Conference on "Economics of Culture, Institutions and Crime" organised by SUS.DIV, FEEM, University of Padua and CEPR, held in Milan on 20-22 January 2010.

(lxxxviii) This paper was presented at the International Workshop on "The Social Dimension of Adaptation to Climate Change", jointly organized by the International Center for Climate Governance, Centro Euro-Mediterraneo per i Cambiamenti Climatici and Fondazione Eni Enrico Mattei, held in Venice, 18-19 February 2010.

(lxxxix) This paper was presented at the 15th Coalition Theory Network Workshop organised by the Groupement de Recherche en Economie Quantitative d'Aix-Marseille, (GREQAM), held in Marseille, France, on June 17-18, 2010.